ENDOGENOUS CONSTITUTIONS*

Davide Ticchi and Andrea Vindigni[†]

First Draft: February 2002. This Draft: September 2005

Abstract

We present a theory of the choice of alternative democratic constitutions, a majoritarian or a consen-

sual one, in an unequal society. A majoritarian democracy redistributes resources from the collectivity

toward relatively few people, and has a relatively small government and low level of taxation. A consen-

sual democracy redistributes resources toward a broader spectrum of social groups but also has a larger

government and a higher level of taxation. We show that a consensual system turns out to be preferred

by society when ex ante income inequality is relatively low, while a majoritarian system is chosen when

income inequality is relatively high. We also obtain that consensual democracies should be expected to

be ruled more often by center-left coalitions while the right should have an advantage in majoritarian

constitutions. The implications for the relationship between inequality and redistribution are discussed.

Historical evidence and a cross-sectional analysis support our results.

Keywords: Endogenous Constitutions, Consensual Democracy, Majoritarian Democracy, Inequality.

JEL Classification Numbers: D31, D72, P16.

*This is a substantially revised and extended version of IIES working paper No. 726, November 2003. We wish to thank

Daron Acemoglu, Alberto Alesina, Roland Bénabou, Walter Garcia-Fontes, José Garcia Montalvo, Edward Glaeser, Gene

Grossman, Arend Lijphart, Massimo Morelli, Elias Papaioannou, Torsten Persson, Gérard Roland, Stefano Sacchi, Gilles Saint-

Paul, Davide Sala, James Snyder, Guido Tabellini, Jonathan Temple, Ernesto Villanueva, two anonymous referee and seminar

participants at Princeton University, Brown University, MIT, Toulouse University, University of Turin, University of Ancona,

IGIER-Bocconi, Yale University, University of Modena and University of Urbino for useful comments. A special thank to

Antonio Ciccone and Howard Rosenthal for long discussions, encouragements, and many valuable comments and suggestions.

Ticchi gratefully acknowledges financial support from the University of Urbino, the University Pompeu Fabra and the European

Commission through the RTN grant "Specialization Versus Diversification." Vindigni gratefully acknowledges financial support

from Princeton University, MIT and IIES. The usual disclaimers apply.

[†]Davide Ticchi: Department of Economics, University of Urbino (Italy) and University Pompeu Fabra (Spain). E-mail:

ticchi@uniurb.it. Andrea Vindigni: Department of Politics, Princeton University and IZA. E-mail: vindigni@princeton.edu.

1

1 Introduction

Democracy is defined as the "government of the people." As Arend Lijphart stresses (e.g. Lijphart, 1999), however, this expression can be understood in two fundamentally different senses, which correspond to the two kinds of democratic political regimes actually observed around the world. "The people" can be taken to mean "the majority of the people" or, alternatively, "as many people as possible." The first sense corresponds, in Lijphart's terminology, to the model of majoritarian democracy (or Westminster model), where political power tends to be concentrated in the hands of a limited number of individuals, and the second sense to the model of consensual democracy, where political power is instead widely shared and dispersed.

Whereas the institutional architectures of the two democratic models differ along several dimensions, perhaps their single most important distinctive feature is the electoral system. Majoritarian democracies are characterized by a majoritarian ("plurality rule" or "first-past-the-post") electoral system, whereas consensual democracies by a proportional one. Moreover, in terms of the executive-legislative relations, the majoritarian model is characterized by the dominance of the former, and the consensual model by a more balanced distribution of political power between the two organs.

A typical example of majoritarian democracy is that observed in the U.K. and other Commonwealth countries, which share the combination of a majoritarian electoral law and of a parliamentary form of government, leading to a substantial concentration of power in the hands of the prime minister. A majoritarian electoral law typically leads to the formation of a two parties system. Moreover, the leader of the largest party is also the head of the government. Vice versa, the combination of a proportional electoral law and of a parliamentary government normally observed in Continental Europe (with the significant exception of France), is a "potent impetus toward consensus democracy" (Lijphart, 1999, p. 303). In fact, a proportional electoral law favors the formation of both a multi-party system and of coalition governments leading to the representation of diverse social interests.

The role of key constitutional norms in shaping fiscal policy outcomes in representative democracies has been emphasized widely in the political economics literature. Important examples include an earlier contribution of Myerson (1993), as well as the more recent ones of Persson, Roland and Tabellini (1997, 2000), Persson and Tabellini (2000), Austen-Smith (2000), Lizzeri and Persico (2001) and Milesi-Ferretti, Perotti and Rostagno (2002). All these papers study the effects of constitutional provisions on some fiscal policy outcomes, but are also all based on the premise of taking political institutions as given.

In this paper we attempt instead to provide an economic theory of the choice of a democratic constitution

on the base of one primitive fundamental: the (pre-tax) distribution of income within a society. Our main result is that a majoritarian constitution is more likely to be chosen when the degree of income inequality is relatively high, while consensual democracy is more likely to arise in relatively homogeneous societies.¹

We present a simple public finance model where fiscal policy is about the provision of some public goods, financed with proportional taxation of income. The public goods considered are local, or group-specific, in the sense that each of them is desired by one and only one of the three social groups (or "classes") which compose the society and that are identified by their level of pre-tax income: the poor, the middle class and the rich. We go on by characterizing the political equilibrium of the model in a majoritarian and in a consensual democracy respectively. A key assumption that we maintain in each constitutional environment is that politicians are citizen-candidates who have a direct interest in the policy implemented and cannot credibly commit to implement any policy different from their preferred one.

We assume that fiscal policy in a majoritarian democracy is decided by a "leader" elected directly by the people through a majority voting process. We demonstrate that in equilibrium the winner is always a rich citizen-candidate as the rich enjoy a natural advantage over the other two classes, arising from the interaction of their relative fiscal conservatism and the majoritarian electoral law, and conclude that the structure of majoritarian democracy biases policy outcomes in favor of this group.

In a consensual democracy fiscal policy is decided by a coalition government formed as the outcome of some legislative bargaining process among the members of a parliament elected with a proportional electoral law. We show that the government coalition depends on the distribution of income. According to our model, in a consensual democracy a middle class and rich (middle class and poor), or center-right (center-left), government coalition is more likely to be formed when the distribution of income is more (less) polarized. We show that taxation and size of government in a consensual democracy under a center-left coalition are higher than under a center-right one. Moreover, taxation and size of government are generally higher in a consensual than in a majoritarian democracy.

Finally, we evaluate from the point of view of the different groups of citizens the welfare implications of the two types of political institutions, and we let individuals vote in an "original position" in absence of any veil of ignorance on which constitution to adopt. We obtain that a society with high income inequality prefers a majoritarian constitution while consensual democracy is preferred when inequality is lower.

¹The focus on the distribution of income is motivated by the insights provided by the positive political economics theory of taxation and redistribution in democracies (e.g. Meltzer and Richard, 1981), which stresses the importance of this variable in shaping fiscal policy outcomes when individual preferences are aggregated directly by majority voting.

We go on to discuss some historical evidence supporting our claim that the key constitutional principles should be interpreted from an economic perspective, namely as reflecting the interests of particular social groups or classes as opposed to the "public good", as well as our key result of the importance of income inequality for institutional choice. For example, we provide evidence that the constitution of the U.S. has been drafted to reflect essentially the interests of the economic elite of the time, and something similar happened in other majoritarian democracies as diverse as the U.K. and Chile. We show that, consistently with the predictions of our model, consensual constitutions have been chosen by several Continental European countries when income inequality in these societies was relatively low and the political voice of the masses was loud. We also present the results of a cross-sectional analysis in a sample of 57 democracies and in a sub-sample of 31 parliamentary democracies. We use the income inequality at the time of (or before) the adoption of the constitution to avoid problems of endogeneity and find that, as predicted by our theory, the degree of inequality is a highly significant determinant of the constitutional choice.

The model has also some other important general implications concerning politico-economic outcomes in representative democracies. A first result, corroborated by the existing empirical evidence, is that taxation and overall size of government in consensual democracies are higher than in majoritarian ones. However, in our model, this is due to a *selection bias* in the composition of the government coalition. Consensual democracies should be expected to be ruled relatively often by center-left coalitions, more willing to tax and redistribute income, while the more fiscally conservative right should have an advantage in majoritarian constitutions. Finally, our constitution selection theory may shed some light on the relationship between inequality and redistribution, suggesting that the former not only affects fiscal policy (in a nonlinear way) for a given constitution, but it also influences the choice of the constitution itself. As a result, the relationship between income inequality and redistribution may well be absent or negative.

Our work is complementary to Alesina and Glaeser (2004). They argue that proportional representation (PR) leads to more redistributive fiscal policies and is thus naturally preferred by the left, whereas the right prefers majoritarian electoral systems because of the conservative fiscal policy outcomes to which these lead. They also present historical evidence showing that the diffusion of PR across Continental Europe in the XX century was related to the political strength enjoyed by the left and by the workers' movement. The political power of the left in the U.S. was instead too low to impose PR. Alesina and Glaeser link the differences in the power of the left across countries to geography, ethnic fragmentation and military success. Our theory of constitutional choice shares with Alesina and Glaeser's the premise that different electoral systems generate

different fiscal policy outcomes and that this is taken into account by agents when choosing the constitution. We also coincide in that upper classes are in favor of majoritarian democracy, and the corollary that older constitutions (as U.S. and U.K.) should be majoritarian as they reflect the greater political voice enjoyed by the elites at the time of their drafting.² The main difference is that we link the constitution selection to income inequality, which allows us to perform a formal statistical analysis of our theory.³ Moreover, our model provides microfoundations in a unified framework for the difference in fiscal policy outcomes across constitutions as well as for the preferences of the various groups for different constitutions.

Our paper is also related to a recent literature that investigates, from different perspectives, the endogenous choice of various political institutions. For example, Acemoglu and Robinson (2000, 2005) have recently explained the transition from oligarchy to democracy as a strategy pursued by the elites to provide redistribution to the poor in order to prevent social unrest. Aghion and Bolton (2003) deal with the normative issue of the choice of an optimal majority rule in an incomplete contracting framework. Aghion, Alesina and Trebbi (2004) analyze the optimal degree of "insulation" of policy-makers. Other contributions along these lines include Barbera and Jackson (2004) and Messner and Polborn (2004).

The paper is organized as follows. Section 2 describes the basic economic setting and the public finance problem we focus on. Sections 3 and 4 deal with the political equilibrium of the model in majoritarian and consensual democracy respectively. Section 5 characterizes the properties of the political equilibrium within and across constitutions. Section 6 deals with the key issue of the endogeneity of the constitution and its relation with the distribution of income. Section 7 presents some pieces of historical evidence supporting our theory. Section 8 shows the results of our cross-sectional analysis on the relationship between income inequality and constitutional choice. In Section 9, we review some empirical studies on the fiscal policy outcomes across constitutions and discuss them from the perspective of our model. Section 10 concludes.

2 A Simple Model of Public Finance: Basic Setup

We consider a simple model of "local" (that is, group specific) public goods provision based on Persson and Tabellini (2000, ch. 7). A society is made up by N > 1 groups of individuals. For convenience, we focus on the case where N = 3. Group $j \in \Im \equiv \{p, b, r\}$ has size (measure) m^j and each individual of that group

²On the other hand, our theory predicts that only the middle class always prefers consensual democracy, while the constitutional preferences of the poor are shaped by the underlying distribution of income.

³ However, we also recognize that groups with more political power did sometime impose their preferences.

has an exogenous pre-tax income equal to y^j . Total population is made by a continuum of unitary measure $\sum_{j\in\Im} m^j = 1$ and with no loss of generality we assume that $\max\{m^p, m^r\} < m^b < \frac{1}{2}$, that $m^i + m^l > \frac{1}{2}$, $\forall (i,l) \subset \wp(\Im)$, the set of all subsets of \Im , and that $y^p < y^b < y^r$. This means that group b is the largest one and has an intermediate level of income, so that it is natural to identify it with the "middle class." Group p and group r correspond vice-versa to the "poor" and to the "rich" people. The absolute majority (or plurality) of votes is reached by the combination of any pair of groups. Notice also that the above assumptions are sufficient to ensure that the voter with median income (i.e. the median voter if preferences are single-crossing in income) belongs to group b. Finally, we assume that $y^b < \overline{y} \equiv \sum_{j \in \Im} m^j y^j$: the voter with median income is poorer then the (virtual) mean voter, which means that the distribution of income is skewed to the left consistently with the empirical evidence.

We assume that the utility function of each member of group j has the following quasi-linear form

$$w^j = c^j + H(g^j) \tag{1}$$

where c^j denotes the consumption of a private good and g^j the level of the type j public good provided. $H(\cdot)$ is a smooth, increasing and concave function and satisfies the Inada conditions.⁴ We also assume that H(0) = 0. The Inada conditions guarantee that at the optimum each group will always strictly prefer to have some taxation and some provision of its desired public good to the alternative of no taxation and no public good. All the above properties are satisfied by the constant elasticity functional form $H(g^j) = A(g^j)^{\alpha}$, where A is a constant and $\alpha \in (0,1)$. At some point we will use such preference specification to obtain some analytical and numerical results.⁵

Each group is perfectly homogeneous. Heterogeneity is only between groups and is related to the differences in the pre-tax income level and to group-specific preferences on the public good to be provided. The specification of preferences in equation (1) implies that each group values one particular public good only (that is, it gets no utility from the provision of any other public good) and there are as many kinds of public goods as the groups of people. One could think to them as pure Samuelsonian (non rival and non excludable) public goods or as publicly provided private goods, like education, health and housing. The first interpretation corresponds somehow to the extreme case of the existence of three pure public goods, on which the different groups of individuals have different preferences. The second interpretation may capture

This means that $H_g(\cdot) > 0$, $H_{gg}(\cdot) < 0$ and $\lim_{g^j \to 0} H_g(g^j) = \infty$.

⁵The quasi-linearity assumption simplifies the analysis but it is not essential. As it will be clear later, what is essential is a preferences specification generating a negative association between the income level and the desired tax rate.

the fact that a significant part of government expenditure is about the provision of private goods and that different income groups may have very different preferences on them.⁶ For example, Besley and Coate (1991) show that, allowing for different quality levels of the public goods, a *de jure* universal provision scheme does not imply that it is *de facto* universal and explain why some publicly provided private goods like health care may go to the advantage of the poor and not to the rich. Fernandez and Rogerson (1995) discuss the case of higher education and emphasize how the public provision of it can benefit higher-income individuals at the expense of the poor.⁷ However, the important feature of our redistribution scheme is the possibility of targeting the benefit of the redistribution toward specific social groups.

Income is taxed at a proportional rate $\tau \in [0,1]$ that will be determined later as a part of the political equilibrium of the model. Therefore, the budget constraint of the agents of group j is simply $c^j = (1-\tau)y^j$. We also assume that the government can finance public expenditures only out of the revenues generated by income taxation. In equilibrium, g^j is positive only when group j is part of the government. If we incorporate this result in the public sector budget constraint, the latter can be rewritten as $\sum_{j\in\Omega} g^j \leqslant \tau \sum_{j\in\Im} m^j y^j = \tau \overline{y}$, where $\Omega \equiv \{j \in \Im : j \text{ is part of the government}\} \subseteq \wp(\Im)$.

In the next two Sections, we derive and characterize the political equilibrium of our model, namely the tax rate τ , the overall level of public expenditure and its composition $G \equiv (g^p, g^b, g^r)$, in the case of both a majoritarian constitution and a consensual one. Since the constitution is at this stage still taken as given, these equilibria can be considered as partial political equilibria. Then, we characterize the general political equilibrium where the constitution will be itself endogenous and chosen by the society.

We assume that *voting* is *sincere* in any constitutional environment⁸ and model the political process going on within a majoritarian or a consensual democracy drawing on the citizen-candidate apparatus of Osborne and Slivinsky (1996) and Besley and Coate (1997).⁹ We adopt a model of endogenous political candidacy since we want to emphasize the link existing between individual preferences (of citizens as well as

⁶In our formulation the public good's per capita consumption of the group does not depend on the size of the group and this might not be always the case if we refer to the provision of private goods. However, one might think that the provision of such goods implies primarily a fixed cost and a low (at the extreme zero) variable cost.

⁷In other words, the specification of preferences in equation (1) could also be interpreted as a reduced form of a more general model where some form of group-specific heterogeneity in preferences regarding the provisions of different public goods arises as an equilibrium outcome, rather than by assumption, as it does in our model.

⁸The sincere voting assumption can be justified in game-theoretic terms by noticing that each individual regards himself as an atomistic subject and therefore considers his vote irrelevant in conditioning the outcome of the elections.

⁹Osborne and Slivinsky (1996) assume sincere voting while in Besley and Coate (1997) individuals are strategic. In this sense our model is closer to the first one.

of politicians) and individual income. Moreover, a key advantage of this model is to allow for the existence of an equilibrium even when individual preferences fail to be single-peaked, and when therefore there may not be an equilibrium under simple majority voting with exogenous candidates. This is a potentially serious problem in our model since the policy space, namely the set $\{(\tau,G) \in [0,1] \times \Re^3_+ : \sum_{j \in \Omega} g^j \leqslant \tau \overline{y}\}$, is not unidimensional and it is well known that political equilibria based on simple majority voting may fail to exist when the social choice process has a multidimensional object. The citizen-candidate model allows us to avoid the problem of non-existence of an equilibrium. At the same time, we are able to show that the main drawback of it, namely the generic multiplicity of political equilibria, is not an issue in our economy.

3 Majoritarian Democracy

We assume that in a majoritarian democracy fiscal policy is decided by a "leader" elected directly by the people through a majority voting process among the menu of citizen-candidates participating to the election. With this assumption, we mean to capture the winner-takes-all nature of political competition going on within a majoritarian democracy, as well as the dominance of the executive power over the legislature, with a modelling framework as simple as possible. A first interpretation of the model is that of a parliamentary democracy with a majoritarian electoral rule. Indeed, if we had assumed the existence of a parliament whose members are elected in single-candidate districts (as in the U.K.), such that the distribution of agents across them is roughly the same as the overall distribution, we would have obtained the same political equilibrium. The model can also be interpreted as a presidential regime with a legislative assembly elected with a majoritarian electoral law (as U.S.). In this case, both organs should be expected to have roughly the same preferences on fiscal matters as they are elected with the same rule, so that independently on the relative power of the two organs these democracies are clearly majoritarian. The power of the president relatively to that of the legislative assembly is important when the latter is elected with PR, which is generally the case in Latin American countries. These countries are generally characterized by a relatively powerful president that plays a very important role in the legislative process and that is often also in a dominant position with

¹⁰Our characterization of majoritarian democracy may appear as too stylized if referred to the American political system, of which it misses important features as the possibility of observing a "divided government" (e.g. Alesina and Rosenthal, 1995, 1996), judicial review and the whole system of checks and balances contemplated by the constitution. A similar assumption is made in Austen-Smith (2000). However, in our view, the main results of the model should be independent from this omission.

respect to the legislature. Therefore, such countries are examples of majoritarian democracies. 11

The menu of candidates is endogenous and one individual runs for office if and only if, in equilibrium, the net gain of doing so (the difference between the utility he gets if does-does not run) exceeds the exogenous cost of running. The winner of the election is the candidate gaining the plurality of votes and he alone decides on fiscal policy. Voting takes place only once.

To characterize the political equilibrium under a majoritarian constitution, it is useful to start from the benchmark case of the unconstrained preferred policy of each social group. Then, suppose that a member of group j (which one is irrelevant given the assumption of perfect within group homogeneity) could act as a dictator and implement his preferred policy ("dictatorial policy"). It is clear that he would not spend anything in any public good other than his preferred one, so that $g^i = 0$, $i \neq j$ and $g^j = \tau \overline{y}$. Hence, he would maximize the following utility $w^j = (1 - \tau) y^j + H(\tau \overline{y})$. The (unique) optimal dictatorial tax rate of group j that solve this problem is 12

$$\tau^{j} = \frac{H_g^{-1}(y^j/\overline{y})}{\overline{y}}. (2)$$

It is straightforward to verify that $\frac{\partial \tau^j}{\partial y^j} < 0$: the richer is a group j member (for a given mean level of income), the higher is the marginal cost of public good provision he faces and the lower is his demand for his preferred public good. Hence, the dictatorial tax rates for the three groups can be ordered as: $\tau^r < \tau^b < \tau^p$.

No commitment technology is assumed to be available, and therefore candidates cannot announce credibly before the election to pursue, if elected, any policy different from their preferred one. This means that in a majoritarian democracy the menu of possible policies is included in the set $\{(\tau^j, G^j)\}_{j \in \mathbb{S}} \subset \{[0, 1] \times \Re^3_+ \times \Omega\}$, where τ^j is defined as in (2), $G^j = (e^j)g^j$, e^j denotes the j^{th} element of the canonical base spanning \Re^3_+ , and $g^j = \tau^j \overline{y}$. Let k denote some private benefit of being in office, which is either a psychological benefit or a non-taxed monetary income, and ε be the cost of running. Both are exogenous and equal for everybody with $k \geqslant \varepsilon$. Now, we can state the main result of this Section, which is contained in Proposition 1.

¹¹Lijphart (1999) and Linz (1994) argue that presidential government as such is inherently an expression of majoritarian democracy as presidential elections are winner-takes-all in nature and all the executive power is concentrated in the hands of one single person (see Shugart and Carey, 1992, for a discussion on this point). In an extension of our model (available from the authors), we show that essentially the same political equilibrium carries over in an institutional environment where the president has relatively large legislative power and the assembly is elected with PR. We therefore conclude that what matters for fiscal policy outcomes in presidential regimes with PR is the electoral law of the president.

 $^{^{12}}$ It is immediate to verify that the second order condition is satisfied. The Inada conditions imposed on $H(\cdot)$ imply that the tax rate is always strictly positive but a tax rate strictly lower than one cannot be excluded.

Proposition 1. The model has a unique political equilibrium with the following features. Only rich citizen-candidates run for office, only the public good preferred by the rich is provided and the tax rate τ is set at the level τ^r preferred by the rich at the level defined by equation (2) for $y^j = y^r$.

Proof. See Appendix.■

Notice that four elements of the model are important for the results of Proposition 1. First, no one group has the majority of the votes alone. Second, the utility function is chosen in such a way that the rich, as dictator, is the group that prefers the lowest taxes. Third, the winner-takes-all nature of the electoral process: in two-candidate contests between the rich and another group, the rich always win since they prefer less taxation.¹³ Fourth, if a group expects to lose an election, no candidate is forthcoming.

4 Consensual Democracy

In a consensual democracy voters do not elect a leader directly but rather elect their representatives to the parliament. We assume the existence of a parliament composed by a continuum of measure $\rho \in (0,1)$ of members which are elected with a pure proportional electoral rule in a single nation-wide electoral district. The government is formed as the outcome of a process of legislative bargaining among the representatives of the different groups and it expresses a certain parliamentary majority.¹⁴ We also assume that the plurality of parliamentary votes is sufficient to form a government.

The policy formation process corresponds to the following three-stages game: 1) the entry of candidates stage; 2) the voting stage; 3) the legislative bargaining stage.

Assuming that there are three groups in the parliament and that no group has the absolute majority of parliamentary members (which will be the case in equilibrium), events take place at the legislative bargaining stage according to the following protocol.

• Round 1 of the bargaining game: the head of the representatives (appointed at random) of the group having the relative majority of seats in the parliament is called to make a policy proposal to the head

¹³This implies that, off-equilibrium, it is possible to observe an "extreme coalition" made up by the rich and the poor. As it is clear from the proof of Proposition 1, in case there are two candidates, a rich and a middle class agent, the poor prefer to vote for the rich since the latter's fiscal conservatism is the best alternative they have.

¹⁴In modelling the legislative bargaining process taking place in a consensual democracy, we partly draw on the agenda-setting model of Romer and Rosenthal (1979), as well as on the model of legislative decision making in a three-party proportional representation system offered by Austen-Smith and Banks (1988).

of the representatives of one other group of his choice. Given that "buying" votes is costly, only two groups coalitions will be observed and a version of Riker's minimum size coalition principle will apply.

- If the proposal is accepted, the government coalition is formed and the agreed policy is implemented.
- Round 2 of the bargaining game: if the proposal is not approved, a second agenda setter is appointed randomly by nature between the representatives of the two groups of which no member was agenda setter at round 1. More precisely, a member of either of these groups is appointed as agenda setter at round 2 with probability equal to the share of the parliamentary seats of his group, relative to the total number of seats of the two groups. Then, the second agenda setter has the opportunity to form a government (just as the first one) and formulates a coalition proposal to one other group of his choice.
- If no proposal is approved at round 2, the game ends and the *status quo* policy is implemented. We assume that the *status quo* policy corresponds to no taxation and no public goods provision.

Notice that our modelling of the policy making process in a consensual democracy is innovative in at least two dimensions. First, we study a legislative bargaining process between citizen-candidates representatives. Second, we analyze how the distribution of income shapes fiscal policy outcomes through the non-standard channel of the bargaining power of the different classes, which is endogenous and turns out to depend on the income distribution itself.

4.1 Entry of Candidates, Voting and Bargaining

The equilibrium of the policy formation game must be sequentially rational, which means that the Nash equilibrium at each stage of the game must rationally anticipate its subsequent equilibrium path.

The assumption of sincere voting and the citizen-candidate structure clearly imply that each individual will simply vote for a candidate from his own social group. This fact and the proportionality of the electoral law imply that group j elects a total of ρm^j representatives.¹⁵ That is, the parliament is a mirror-image of society in the sense that the distribution of seats across the three groups exactly reflects the distribution of the population across these groups.

¹⁵ It is straightforward to verify that the supply of candidates from each group is not lower than ρm^j . Since the parliament is large, the policy outcome does not depend on whether any individual does or does not run for office. Therefore, a citizencandidate of group j runs for office if and only if $pk \ge \varepsilon$, where p is the probability of being elected, pinned down at $p = \frac{\varepsilon}{k} \in (0, 1]$ by the assumption of free entry of candidates.

The agenda setter at round 1 is a representative of the middle-class, which (being the largest class) has the largest number of seats in the parliament. Moreover, if the middle class fails to form a government, the second agenda setter is appointed randomly by nature and chosen between the representatives of the poor and the rich. By assumption, the probability that a poor (rich) will be the agenda setter at the second round (conditional of the game reaching it) is equal to the share of the seats of the poor (rich) of the combined number of seats of the poor and of the rich. Hence, $\phi = \frac{\rho m^p}{\rho m^p + \rho m^r} = \frac{m^p}{m^p + m^r}$ is the probability that a poor is appointed as agenda setter at round 2. Conversely, the probability that the agenda setter at the second round is a rich is equal to the complementary probability $(1 - \phi)$.

It is clear that ϕ can be interpreted as an index of the bargaining power of the poor: the higher is the number of the poor m^p , the higher is ϕ , the higher is the probability that the poor are agenda setter at the second round, the higher is their expected utility at that stage of the game, and therefore the higher will be the public good that the middle class agenda setter (at round 1) provides them for any given level of taxation so to accept her government coalition proposal. Moreover, as we will show later, ϕ is also a measure of income inequality: other things equal, a higher ϕ corresponds to a more unequal income distribution.¹⁶

The legislative bargaining game has a unique (subgame-perfect) Nash equilibrium. The first agenda setter (from the middle class) formulates a coalition formation proposal based on a fiscal policy program to one other group only, given that no more than that is needed to reach a parliamentary majority. The coalition formation offer leaves the group receiving it indifferent between accepting and rejecting it, namely gives to the recipient a level of utility equal to its outside option and the offer is accepted. Therefore, the question we need to answer is: which group (among the poor and the rich) is the cheapest to buy? To answer this question, we first solve our bargaining game by backward induction starting from the second round.

We denote the group of the agenda setter with h, the other group part of the government with l and the stage of the game with s. Therefore, $\tau_{s,h,l}$ is the tax rate proposed to group l by the agenda setter h at round s of the game. The correspondent level of public good received by the group i will be $g_{s,h,l}^i$. Similarly, the level of utility of the group i is $w_{s,h,l}^i$.

¹⁶ Finally notice that the probability $\phi \in (0,1)$ because m^p and m^r are always positive. However, in the analysis presented below we also consider the limit cases where $\phi = 0$ and $\phi = 1$ because they allows us to define the tax rates and utilities in the all range [0,1].

4.1.1 Round 2 of the Bargaining Game

Lemma 1. At round 2 of the bargaining game, the poor are always part of the government coalition; the middle class is so only if the agenda setter is a poor and the rich only if the agenda setter is a rich.

Proof. At round 2, the outside option of each group is its *status quo* utility, i.e. its gross income. Since the agenda setter optimizes giving to the coalition partner what is strictly necessary to induce it to accept the policy proposed, the policy menu $(\tau_{2,h,l}, g_{s,h,l}^l)$ offered from the agenda setter h to group l satisfies the condition $(1 - \tau_{2,h,l}) y^l + H(g_{2,h,l}^l) = y^l$. Consider the schedule $g_{2,h,l}^l \equiv g_{2,h,l}^l(\tau_{2,h,l}; y^l)$ defined implicitly by this equation. Holding constant $\tau_{2,h,l}$, this schedule is such that $\frac{\partial g_{2,h,l}^l}{\partial y^l} = \frac{\tau_{2,h,l}}{H_g(g_{2,h,l}^l)} > 0$. This means that the richer is a group, the more it has to be compensated in terms of public good provision for any level of taxation. Thus, if the rich is appointed agenda setter at the second round, he will always prefer the poor to the middle class as coalition partner. Alternatively, if the second round agenda setter is poor, the middle class will be cheaper to buy than the rich. Finally, the assumptions on the protocol of the legislative bargaining game imply that the middle class is never the agenda setter at round 2.

Therefore, if at round 2 the agenda setter is a rich, then by Lemma 1 the poor will be the coalition partner. The substitution of the equilibrium government budget constraint $(\tau_{2,r,p}\overline{y}=g_{2,r,p}^p+g_{2,r,p}^r)$ in the utility function of the rich implies that their maximization problem is

$$\max_{\{\tau_{2,r,p}\}} w_{2,r,p}^r = (1 - \tau_{2,r,p}) y^r + H(\tau_{2,r,p} \overline{y} - g_{2,r,p}^p)$$
(3)

subject the participation constrain of the poor $(1 - \tau_{2,r,p}) y^p + H(g_{2,r,p}^p) = y^p$. If at round 2 the agenda setter is a poor, he makes a coalition with the middle class and the maximization problem is derived in a similar way. More details on the optimization problems at round 2 and at round 1 can be found in Ticchi and Vindigni (2003).

4.1.2 Round 1 of the Bargaining Game

While the poor are always part of the government coalition if the game reaches round 2 (an off-equilibrium event), this does not need be the case at round 1. At this stage of the game, the middle class agenda setter will form the government coalition with the group that allows her to reach the highest level of utility from the implemented policy. This policy will be such to leave the group receiving the offer just indifferent between accepting it and going to the second round. As we will see, the expected utility of each group at

round 2 depends positively on its probability of being agenda setter at that stage. Hence, the higher is the probability ϕ of the poor of being agenda setter at round 2, the higher is their expected utility at this stage of the game, the more costly is for the middle class to buy their vote at round 1, and therefore the less likely is that they are part of the government coalition. In what follows, we will establish a global result which identifies the winning coalition in terms of a critical value of ϕ . To proceed in this direction, we first define the maximization problems of the middle class under the two possible coalitions.

If the government coalition is made up by the middle class and the rich, the participation constraint of the rich at round 1 is

$$(1 - \tau_{1,b,r}) y^r + H(g_{1,b,r}^r) \geqslant (1 - \phi) \left[(1 - \tau_{2,r,p}) y^r + H(g_{2,r,p}^r) \right] + \phi \left(1 - \tau_{2,p,b} \right) y^r. \tag{4}$$

The left hand side of (4) represents the utility of the rich if the middle class' policy proposal at round 1 is implemented, while the right hand side is their expected utility conditional on the game reaching round 2.

Substituting the equilibrium government budget constraint $(\tau_{1,b,r}\overline{y}=g_{1,b,r}^b+g_{1,b,r}^r)$ in the utility function of the middle class, their maximization problem becomes

$$\max_{\{\tau_{1,b,r}\}} w_{1,b,r}^b = (1 - \tau_{1,b,r}) y^b + H(\tau_{1,b,r} \overline{y} - g_{1,b,r}^r)$$
(5)

subject to the participation constraint of the rich (4). This constraint may not be always binding however. This is the case when the dictatorial policy of the middle class gives to the rich a higher utility than their expected utility at round 2. In this situation the consensual democracy equilibrium is equivalent to the dictatorship of the middle class which obtains the maximum level of utility by implementing her unconstrained preferred policy.¹⁷ When the participation constraint of the rich is binding, the first order condition relative to the maximization problem (5) is

$$y^{b} = H_{g}(\tau_{1,b,r}\overline{y} - g_{1,b,r}^{r}) \left(\overline{y} - \frac{y^{r}}{H_{g}(g_{1,b,r}^{r})}\right)$$
(6)

¹⁷ From an inspection of (4), it is immediate to verify that this is always the case whenever, as ϕ approaches one, the middle class dictatorial tax rate τ^b is lower than $\tau_{2,p,b}$. For example, it is easy to prove analytically that $\tau^b < \tau_{2,p,b}$ at $\phi = 1$ if $y^p = 0$. Indeed, under these conditions $\tau_{2,p,b}$ is defined by the equation $\frac{y^b}{H_g(\tau_{2,p,b}y^b)} = \overline{y}$ and the dictatorial tax rate of the middle class by $\frac{y^b}{H_g(\tau^b\overline{y})} = \overline{y}$, which imply that $\tau_{2,p,b}y^b = \tau^b\overline{y}$. From $y^b < \overline{y}$ follows that $\tau^b < \tau_{2,p,b}$. Our numerical simulations, that we discuss in the next Section, show that this result also holds for values of ϕ different from one or when the income of the poor is positive. Clearly, the lower is the income of the poor and the higher is $\tau_{2,p,b}$, which in turn makes more likely that the participation constraint of the rich is not binding.

and this equation allows us to get $\tau_{1,b,r}$ and $g_{1,b,r}^r$ if combined with equation (4). Then, $g_{1,b,r}^b$ is obtained from the government budget constraint.

If the coalition government is made up by the middle class and the poor, the participation constraint of the poor at round 1 is

$$(1 - \tau_{1,b,p}) y^p + H(g_{1,b,p}^p) \geqslant (1 - \phi) y^p + \phi \left[(1 - \tau_{2,p,b}) y^p + H(g_{2,p,b}^p) \right]. \tag{7}$$

The left hand side of (7) is the utility of the poor at round 1 if the middle class' policy proposal is implemented, while the right hand side corresponds to their expected utility if the game reaches the second round. This participation constraint is always binding (if $y^p > 0$), and therefore hold with the equality sign, because the expected utility of the poor at round 2 is at least equal to their level of income (i.e. what they get if the status quo policy is implemented) given that they are always part of the government coalition at this stage.

The substitution of the equilibrium government budget constraint $(\tau_{1,b,p}\overline{y}=g_{1,b,p}^b+g_{1,b,p}^p)$ in the utility function of the middle class implies that their maximization problem can be written as

$$\max_{\{\tau_{1,b,p}\}} w_{1,b,p}^b = (1 - \tau_{1,b,p}) y^b + H(\tau_{1,b,p} \overline{y} - g_{1,b,p}^p)$$
(8)

subject to the participation constraint of the poor (7). The first order condition of this problem reads

$$y^{b} = H_{g}(\tau_{1,b,p}\overline{y} - g_{1,b,p}^{p}) \left(\overline{y} - \frac{y^{p}}{H_{g}(g_{1,b,p}^{p})}\right). \tag{9}$$

From (9) and (7) we obtain $\tau_{1,b,p}$ and $g_{1,b,p}^p$, while $g_{1,b,p}^b$ is derived from the government budget constraint.

The next Proposition characterizes the outcome of the coalition formation process at the first round of the legislative bargaining game.

Proposition 2. There exists a threshold value of ϕ , $\phi^* \in (0,1)$ such that

- (i) if $\phi < \phi^*$, the government coalition is made by the middle class and by the poor;
- (ii) if $\phi > \phi^*$, the government coalition is made by the middle class and by the rich.

Proof. See Appendix.■

The intuition for this result is straightforward. When ϕ is relatively low ($\phi < \phi^*$), the probability that the poor are agenda setter at round 2 is also low, and so is their expected utility at the second round. This means that their vote is relatively cheap to buy at round 1. On the other hand, when ϕ is relatively small,

 $1-\phi$ is relatively high, and so is the probability of the rich of being agenda setter at round 2, which in turn implies that their expected utility at the second round is high and their vote is costly to buy at round 1. Therefore, there exists a level of ϕ sufficiently small that the middle class prefers to make a government coalition with poor because their vote is cheaper to buy (than that of the rich). Clearly, the opposite is true when ϕ is relatively high ($\phi > \phi^*$). At the threshold ϕ^* , the middle class is just indifferent between forming a coalition with the rich or with the poor.¹⁸

It is worth noting that this result is independent on the status quo policy assumed. It is clear that changing the status quo may imply a variation in the government coalitions observed at round 2 with the poor not being always the cheapest to buy. However, what is key for our results is that the expected utility of poor and rich at round 2 is increasing in their probability of being agenda setter at that stage of the game. And this is always the case because the utility of any group (poor or rich) when it is agenda setter is higher than when it is not (independently that it belongs or not to the ruling coalition). The status quo may affect the absolute level of the expected utility of each class at round 2 for a given level of ϕ and, therefore, the level of the threshold ϕ^* . However, the middle class will prefer the poor to the rich when ϕ is lower than this threshold, and vice versa.

5 The Size of Government Across Constitutions and Coalitions

While the tax rate chosen under a majoritarian constitution is only a function of the income of the rich (relative to the average one), the tax rate in a consensual democracy under the two possible coalitions is a function of the income distribution, i.e. of both the incomes of the classes $(y^p, y^b, \overline{y}, y^r)$ and the value of ϕ . A comparison of these tax rates is not straightforward due to the strong non-linearity present in the first order conditions defining them. However, by making some assumptions on the levels of income of the classes and the utility function of the individuals we can state the results presented and discussed below. First, we assume a power function specification for the utility derived from the public good: $H(g^j) = A(g^j)^{\alpha}$, with $\alpha \in (0,1)$ and A > 0.

Result 1. If the income of the poor is equal to zero and the income of the rich is sufficiently high a sufficiently high the result in Proposition 2 can be regarded as an application of the general principle by which, in coalition formation games, it can be in fact advantageous to be in a relatively weak bargaining position as that increases the likelihood of becoming a member of the coalition. This is much to the contrary of what happens in a Nash bargaining process, where a lower bargaining power only reduces the share of the surplus of which one can appropriate.

relative to the average income, then the level of taxation and public expenditure (and the size of government) obtaining in a consensual democracy under a government coalition made by the middle class and the poor is always higher than that obtaining under a middle class and rich coalition, which in turn is higher than the taxation and public expenditure obtaining in a majoritarian democracy. In other words, for all values of ϕ the following inequalities hold: $\tau^r < \tau_{1,b,r} < \tau_{1,b,p}$.

Proof. See Appendix.■

Without the above assumptions it is not possible to derive further analytical results. Therefore, we have run several numerical simulations whose results are discussed below and that can be found in Ticchi and Vindigni (2003). The simulations show that the two assumptions on the income of the poor and the rich, which guarantee that Result 1 holds and can be proved analytically, are not necessary. Nevertheless, these assumptions provide an insight on the characteristics of the income distribution that lead to that result. In particular, the result that $\tau^r < \tau_{1,b,r} < \tau_{1,b,p}$ for all ϕ is easy to obtain when the income of the poor y^p and the income of the rich y^r are respectively low and high with respect to the average income \overline{y} or, in other words, when there is enough dispersion in the income of the three classes.

From the numerical analysis we have obtained two interesting results. First, a level of y^r sufficiently high relative to \overline{y} is enough to obtain the taxation ranking of Result 1 ($\tau^r < \tau_{1,b,r} < \tau_{1,b,p}$) even when y^p and y^b are both very close to \overline{y} . Second, with an extremely equal income distribution, taxation in majoritarian democracy is always higher than taxation in consensual democracy (regardless of the ruling coalition). These results can be illustrated using two parameterizations for the income of the three classes.¹⁹ The first is: $y^p = 0.9$, $y^b = 0.95$, $\overline{y} = 1$, $y^r = 1.1$. In the second one we just change the income of the rich and use $y^r = 4$. The first parameterization corresponds to a very equal society given that the income of the poor is only 10% lower than the average income while the income of the rich is only 10% higher than the mean.²⁰ In this case the tax rate set in consensual democracy is lower than the tax rate set in majoritarian democracy. The increase in the income of the rich from 1.1 to 4 leads to a reduction in the tax rate set in majoritarian democracy higher than the reduction in the tax rate of the middle class and rich coalition up to the point that the ranking of tax rates of Result 1 generally holds.²¹ The explanation for this result is the following.

¹⁹In the numerical simulations we have used the power function specification for $H(\cdot)$ with A=1 and $\alpha=0.5$.

²⁰In this case the variation in the size of the classes, and therefore of ϕ , cannot change the fact that the distribution is very equal even when there many poor (ϕ high).

Notice that the tax rate of the middle class and poor coalition is not affected by the income of the rich y^r .

The fiscal policy of a single group government should involve, other things equal (i.e. if all groups have a similar income as it is in the first parameterization), a higher tax rate and total expenditure than the policy of a two groups government coalition regardless of how the tax revenues are divided among the public goods provided. Indeed, recall that a public good is provided only if the group which likes it is part of the government coalition and the optimal tax rate of a group is such that the marginal cost of taxation equals the marginal benefit from the public good provision. While the marginal cost of taxation is independent on the number of public goods that are financed with the tax revenues (i.e. the number of groups in the government coalition), the marginal benefit from that increase in taxation decreases with the number of public goods among which this increase in taxation is split. This implies that the tax rate should decrease with the number of groups in the government coalition, which in turn implies that, other things equal, the tax rate in majoritarian democracy should be higher than tax rate in consensual democracy.

However, we now need to explain why Result 1 and most numerical simulations lead to the opposite result, namely that generally the tax rate in consensual democracy is higher than the tax rate in majoritarian democracy. This result is due to the fact that in majoritarian democracy fiscal policy is decided by the group with the highest level of income (the rich), while in consensual democracy fiscal policy is chosen by a government coalition representing two groups with an average level of income lower than the income of the rich. Similarly, the middle class and poor coalition taxes and spends more than the middle class and rich one exactly because it contains a group (the poor) with lower income. If the incomes of the three classes are sufficiently spread, then the latter effect more than compensate the effect (described above) generated by the number of groups in the government leading to the taxation ranking of Result 1 ($\tau^r < \tau_{1,b,r} < \tau_{1,b,p}$).

There are also other features of the relationship between the tax rates across constitutions for different income distributions that deserve to be more deeply analyzed. Our numerical simulations have shown that taxation in majoritarian democracy is lower than taxation under the middle class and poor coalition even if the dispersion in the income levels of the three classes is very small.²² A slightly higher spread in this distribution is necessary if we want the tax rate of the middle class and rich coalition to be higher than the tax rate in majoritarian democracy for all $\phi > \phi^*$. As it will be clear in the next Section, we are interested in the case where $\tau^r < \tau_{1,b,r}$ when $\phi > \phi^*$ and the numerical simulations suggest that this result holds if there is a minimum degree of dispersion in the income levels of the three groups.²³

²² For example, with the above parameterization $(y^p = 0.9, y^b = 0.95, \overline{y} = 1)$ an income level of the rich higher than 1.3 is enough to obtain $\tau^r < \tau_{1,b,p}$ for all $\phi \in (0,1)$.

²³Indeed, using the previous parametrization ($y^p = 0.9, y^b = 0.95, \overline{y} = 1$) and increasing the income of the rich to 1.6 is

Finally, we want to discuss how the tax rates and the value of ϕ^* change as the incomes of the three classes vary. An increase in the income of the rich always leads to a lower tax rate in majoritarian democracy while the tax rate under the middle class and poor coalition is not affected by a variation in y^r . For certain parameterizations, the tax rate under the middle class and poor coalition generally decreases for all $\phi \in (0,1)$ as y^r increases, while decreases for low values of ϕ and increases for ϕ sufficiently high for other parameterizations. However, an increase in y^r always reduces ϕ^* . A decrease in the income of the middle class lowers their utility under both coalitions and therefore ϕ^* can both increase or decrease. As expected, the reduction in the income of the middle class increases taxation in consensual democracy under both coalitions, while taxation in majoritarian democracy is obviously not affected by a variation of y^b . A lower income of the poor increases the utility of the middle class under both coalitions and the sign of the variation of ϕ^* cannot be uniquely determined. Similarly to a reduction in y^b , a lower income of the poor increases taxation in consensual democracy under both coalitions.

6 Income Inequality and Constitutional Choice

Having characterized the political equilibrium of the model under the two possible institutional arrangements, we now turn to the question of which of them would be chosen ex ante by society, when the constitution is endogenous. We assume that the process of constitutional choice takes place in an original position where individuals know their class-status and preferences and before any other politico-economic interaction. We also assume that the decision is taken by simple majority voting, and that the available alternatives are the two constitutional regimes we have considered. Given that there is no veil of ignorance and uncertainty, individuals correctly anticipate what their level of utility would be under the two possible constitutions, and vote consequently. Finally, we assume that there is a minimum degree of dispersion in the income levels of the three groups so that the tax rates ranking of Result 1 holds, or at least that the tax rate set under the middle class and rich coalition is higher than the tax rate under majoritarian democracy when $\phi > \phi^*$.

Proposition 3. If $\phi < \phi^*$ society chooses consensual democracy while it prefers majoritarian democracy when $\phi > \phi^*$.

Proof. It is clear that for the rich and the middle class the constitutional choice has a trivial, albeit opposite, solution. Since majoritarian democracy expresses the dictatorship of the rich, they will prefer it enough to have $\tau^r < \tau_{1,b,r}$ for all $\phi > \phi^*$ and a further increase in y^r implies that $\tau^r < \tau_{1,b,r}$ also for some values of $\phi < \phi^*$.

unconditionally. Similarly, given that the middle class has the relative majority of votes, which allows her to be the first agenda setter in the legislative bargaining game, she will prefer unconditionally the consensual constitution.²⁴ The most interesting decision is the one of the poor, who turn out to be the swing voters. The poor do gain from the higher political inclusion which is typical of consensual democracy, only if they are part of the government coalition as partner of the middle class agenda setter. In this case they are clearly better-off than they are in a majoritarian setting.²⁵ However, we know that this needs not to be always the case, since the ruling coalition does not include them whenever $\phi \in (\phi^*, 1)$. In this instance the poor are actually worse-off in a consensual democracy: they do pay higher taxes (see the discussion in the previous Section) but get as well no provision at all of their specific public good. Therefore, if $\phi \in (0, \phi^*)$ the majority prefers consensual democracy while majoritarian democracy is chosen when $\phi \in (\phi^*, 1)$.

We now present two mean preserving spread of the distribution of income which show that an increase in income inequality makes more likely the adoption of a majoritarian democracy.

Mean Preserving Spread 1. We first consider a transformation of the income distribution that affects the size of the three classes. We suppose that m^p and m^r increase and that m^b decreases in such a way that both the size of the population and the average level of income \overline{y} remain constant. Then, society has a smaller middle class, more rich and more poor; that is, it is more unequal.

We now show that, whilst the threshold ϕ^* is not affected by this transformation, the value of ϕ necessarily increases. Therefore, it becomes more likely that ϕ belongs to the interval $(\phi^*, 1)$ and that a majoritarian democracy is chosen. To this end, let us consider the definition of the average income, $\overline{y} = m^p y^p + m^b y^b + m^r y^r$, and divide both sides by $(1-m^b) = (m^p + m^r)$ taking into account that $\phi = \frac{m^p}{m^p + m^r}$. Then, rearranging terms we get that

$$y^r - \phi(y^r - y^p) = \overline{y} + \frac{m^b}{1 - m^b} (\overline{y} - y^b). \tag{10}$$

A decrease in m^b implies a reduction in the right hand side of (10) as $\overline{y} > y^b$. Therefore, ϕ must increase given that $y^r > y^p$. This means that ϕ can also be interpreted as a measure of income inequality. Again, our

²⁴It is trivial to deduce that the middle class is always better-off in a consensual democracy than in a majoritarian one. Indeed, notice that in consensual democracy the middle class would always have the option of replicating the majoritarian outcome by offering to the rich of forming a coalition and implementing their own preferred policy.

²⁵ If the poor are part of the government coalition they get a utility at least as high as their income level, while in majoritarian democracy they always get a lower level of utility because of positive taxation and no provision of their public good.

model predicts that when income inequality is relatively low ($\phi < \phi^*$) society prefers a consensual democracy, while it chooses a majoritarian system when income inequality is relatively high ($\phi > \phi^*$).

Mean Preserving Spread 2. Another mean preserving spread we consider is generated by a transformation of both the size of the classes and the income of the rich. In particular, we analyze the effect on constitutional choice of an increase in inequality caused by the increase in the income of the rich, accompanied by an equi-proportional increase in the number of the rich and of the poor (and a reduction of the middle class) so that both ϕ and \overline{y} are unaffected.

Given that ϕ remains constant, we need to determine how ϕ^* changes so to understand which constitution is more likely to be chosen by the society. To this end, we need to find out how the two schedules representing the utility of the middle class under the two possible coalitions vary as y^r increases. First, observe that the utility of the middle class in the coalition with the poor $w^b_{1,b,p}$ does not depend on the income of the rich y^r and therefore it is not affected by any variation of it. The utility of the middle class in the government coalition with rich $w^b_{1,b,r}$ is instead a function of y^r . We are not able to show analytically how this schedule changes with the income of the rich but the numerical simulations we discussed in the previous Section point out that it shifts upward (i.e. $w^b_{1,b,r}$ increases) as y^r goes up leading to a reduction in ϕ^* .²⁶ This means that this increase in income inequality leads to a reduction of the range $(0, \phi^*)$ where consensual democracy is chosen, and therefore it makes more likely the adoption of a majoritarian constitution.²⁷

At this point, two issues are worth mentioning. The first is that the model is robust to the removal of some assumptions. For example, the introduction of a general public good would not change our conclusions as long as the taste of the agents for this good is not too high relative to that of the specific public good.²⁸ Assuming a purely deterministic agenda setting scheme in consensual democracy does not affect the results either. Details of these extensions of the baseline framework can be found in Ticchi and Vindigni (2003).

The second issue is about the potential dynamic stability of our model. Although the model is purely static, it contains forces that would make it dynamically stable. If income inequality is relatively high, society should choose a majoritarian constitution. The fiscal policy in this constitutional system should favor the rich and therefore not reduce (or even increase) the initial degree of inequality. Then, society should continue

²⁶More precisely, it is possible to observe for certain parameterizations that $w_{1,b,r}^b$ shifts upward for almost all ϕ except for values close to one where it shift down. However, this is always irrelevant because it happens for values of ϕ very far from ϕ^* .

²⁷This particular mean preserving spread corresponds to the shocks driving, according to our theory, the cases of constitutional change in France, Germany and Italy that we discuss next.

²⁸A similar result obtains assuming that the poor derive some positive utility also from the public good of the middle class.

to prefer a majoritarian constitution. Conversely, if income inequality is relatively low, society will prefer a consensual democracy under which fiscal policy generally reflects the preferences of middle class and poor. This should lead to the provision of public goods that reduce or maintain the initial level of inequality and, therefore, a majority for a consensual constitution.

7 Some Historical Evidence on Constitutional Choice

In this Section we present some pieces of historical evidence on the making of the oldest majoritarian constitutions, like the one of U.S. and the U.K., and the more recent case of Chile. Then, we analyze the adoption of consensual constitutions in Continental European countries in the XX century.

7.1 The Making of Majoritarian Constitutions

The U.S. have the oldest written constitution of the world, dating to 1787. It was drafted by a Constitutional Convention of delegates from all States (with the exception of Rhode Island). Apart from some relatively minor changes, it has remained essentially the same up today. An economic analysis of the American constitution has been provided by Beard (1913), which shares with ours the basic premise that key constitutional principles ought to be interpreted as reflecting the interests of particular social groups or classes as opposed to the "public good". ²⁹ In particular, Beard points out that the economic interests of the members of the Convention essentially corresponded only to those of the rich (the commercial and financial elite as well as of the landlords), who were concerned of securing individual property rights and of guaranteeing the best possible institutional framework for private economic activity. Vice versa, the middle and the lower classes had very little if any voice at all in the Convention, due to both the strong franchise restriction of the time and the low "class-consciousness" of many of the less well-off among the enfranchised. Beard also demonstrates the extraordinary awareness of the economic elite, and of its political and intellectual leadership, about the nature of her interests in the process of constitution making.³⁰ Remarkably, in the tenth number of The Federalist, Madison argues that "The first object of government is the protection of the diversity in the faculties of men, from which the rights of property originate" and that this requires the creation of an institutional framework preventing the exploitation of the rich by the lower classes.

²⁹We thank Alberto Alesina for bringing this book to our attention.

³⁰This also confirms that our assumption on the absence of a veil of ignorance in choosing the constitution is appropriate.

England has the oldest unwritten constitution of the world. It consists in a collection of different documents including the *Magna Charta* of 1215, the *Bill of Rights* of 1689, commonly observed practices and conceptions, as well as some laws. Despite experiencing significant transformations over time, including a gradual extension of franchise since the first decades of the XIX century (1832, 1867 and 1884), the constitution of England always remained majoritarian. The making of the English constitution, like that of the U.S., occurred over a period of time during which the political voice of the upper classes was overwhelming, and our model predicts that the rich always opt for a majoritarian democracy.

The rich have occasionally been able to impose their constitutional preferences even in much more recent times and, just as our theory suggest, they have opted for the majoritarian model. A notable example is the majoritarian constitution of Chile implemented under the leadership of general Pinochet, who arguably represented the political and economic interests of the elites. A politico-economic analysis of this constitution has been offered by Baldez and Carey (1999), who assert (at p. 52) that "The Chilean military regime consciously crafted the 1980 Constitution, the budget process, and the electoral system in order to constrain the policy outcomes generated by elected civilian politicians. To a certain extent, the institutional engineers appear to have succeeded-particularly in designing an electoral system that has disproportionately rewarded the right and a budget process that generally encourages fiscal austerity."

7.2 The Birth of Consensual Democracies

Essential elements of the model of consensual democracy, and in particular a proportional electoral system, have been first introduced in Scandinavian countries (Denmark, Norway, Sweden, Finland) and elsewhere in Northern Europe (the Netherlands) between the 19^{th} century and the beginning of the 20^{th} century. By 1921, all Scandinavian countries had adopted some form of proportional system and none of them has been discarded afterwards, even for a short period (see for instance Lakeman and Lambert, 1963).

The first national election based on some form of PR took place in Denmark in 1856, and the degree of proportionality of the Danish electoral system has been increased gradually until 1920. The evidence on income distribution in Denmark presented in Morrisson (2000) clearly suggests that during the same period the degree of income inequality was overall relatively low and decreasing over time.³¹

³¹The data on income inequality in Denmark are based on the maximum equalization coefficient (MEC), which indicates the share of total income which has to be transferred from the population with income above the average to the other people in order to achieve an equal distribution. This index falls sharply from 0.50 to 0.35 between 1870 and 1900 and was 0.36 in 1925.

Norway introduced PR in 1921, when the degree of income inequality was relatively low. The available evidence illustrated in Morrisson (see Table 2, p. 224) shows a fall in the Gini coefficient from 0.68 in 1855 to 0.49 of the period 1865-1900 to 0.40 of 1900-1910 to 0.34 of 1920.

The Netherlands adopted a proportional electoral law as early as 1917. While no precise evidence at the time of the adoption of the constitution is available, income inequality seems to have generally decreased in the Netherlands since the end of the nineteenth century (Morrisson, 2000).

While the political development of Denmark, Norway and the Netherlands appear to be consistent with our theory, that of Finland and Sweden are less so. Sweden switched to PR in 1907 and, according to Morrisson, inequality has increased in Sweden since 1870, and has had a declining trend from 1914 to 1970, when it has stabilized. Finland adopted PR in 1906, a period where there is limited information on income inequality of this country. However, Morrisson (p. 228) reports that inequality in Finland has increased from 1881 to 1890 and decreased in the twentieth century.

It is worth noting that in all the above countries there is clear evidence that income inequality has declined after the adoption of PR, which suggests that a consensual constitution should help reducing over time the initial degree of income inequality as predicted by our model.

Piketty (2003) documents the occurrence in France during the period 1914-1945 and mostly during World War II, of a sharp fall in income inequality associated to a sensible drop in the top percentile income share. This trend has been driven primarily by the shocks represented by the two World Wars and by the events of the inter-wars period (inflation and Great Depression), which roughly correspond to a change in the income distribution equivalent to our mean preserving spread 2, and which we expect to lead to the adoption of a consensual constitution. Interestingly, in the immediate aftermath of both the First and the Second World War, France switched to PR. In 1919, the Third Republic did so, even if the proportionality of the electoral system was quite weakened. This system was abandoned after the 1924 election, when France reverted to the second ballot in single-member constituencies used until the war. In October 1945, a real proportional system was applied to the election of the first Constituent Assembly. The same system was applied for the election of the Second Constituent Assembly in June 1946 and of the National Assembly in November.

The two World Wars (especially the second one) have presumably affected, in terms of capital disruption and reduction of inequality, other Continental European countries as Italy and Germany. Italy had adopted a majoritarian electoral law at the time of the unification, and switched to PR in 1919, used until the rise of fascism. A proportional system was again adopted for the parliamentary democracy established with the

constitution of 1948. PR has also been adopted in Germany with the constitution of the Weimar Republic in 1919 and reintroduced after 1945. The available evidence (see Morrisson, p. 232) indicates that income inequality has strongly decreased in Germany from 1913 to 1926 and it was low after the Second World War.

As discussed in the Introduction, Alesina and Glaeser (2004) present evidence that the adoption of PR in several European countries reflected the political power of the left. In our view, their finding is in agreement with the predictions of our model as we obtain that the left is in favor of PR when income inequality is not too high. This appears to have been the case at the time of constitutional choice for most countries discussed above.³²

8 Empirical Evidence

The theory presented in this paper has many interesting predictions, but the main result is that income inequality has an important effect on the choice of the constitution. More unequal countries are expected to choose a majoritarian democracy while equal societies should prefer a consensual constitution.

We test this prediction by analyzing the relationship between the type of constitution adopted and the income inequality of the country at the time, or before, the constitution was chosen in order to avoid problems of endogeneity. Given that constitutional reforms are rare (a feature consistent with our model), we perform a cross-sectional analysis. To carry out this empirical study, we make use of the dataset compiled by Persson and Tabellini (2003) (PT from now on) to analyze the economic effects of constitutions.³³ As a measure of income inequality, we employ the Gini index and use the dataset compiled by Deininger and Squire (1996).

PT report the variables maj and pres that define the electoral system and the form of government of each country. In particular, maj is a dummy variable equal to 1 if all the lower house is elected under plurality rule and 0 under a proportional (or mixed) electoral system, while pres is a dummy variable for the form of government, equal to 1 in presidential regimes and 0 in parliamentary democracies. PT classify as presidential only those regimes where the confidence of the assembly is not necessary for the executive and, therefore, most semi-presidential and premier-presidential systems are classified as parliamentary.³⁴

³² Alesina and Glaeser (2004) also discuss the case of Belgium, Switzerland and Austria. We do not because we could not find data on income inequality in these countries for the period when the choice of the constitution was made.

³³The dataset used in their cross-sectional analysis contains 85 countries that have been selected on the base that they have an average *gastil* score lower than 5 for the period 1990-98, i.e. that they can be classified as free or partly free democracies.

³⁴For example, France and Portugal are coded as parliamentary because the legislature in these countries has an exclusive and unrestricted right of censure of the executive. Moreover, the president nominates the government or has an influence over

PT classify the electoral rule and the form of government considering the countries at the beginning of the 1990s. We have taken this classification and identified the first year when this constitution (represented by maj and pres) was first introduced and summarized it in the variable yearcons. Then, we have selected a Gini coefficient for each country from the Deininger and Squire dataset according to the following rules. We have first considered the high quality data and selected the Gini coefficient correspondent to the year when the constitution was introduced, which is given by the variable yearcons. If there was not a data in that year, we have gone backward and selected the first Gini coefficient available. When there was no data before the year of the constitution, we have taken the first Gini coefficient available after that year with the constraint that it was not more than 5 years older.³⁵ Following this procedure, we have obtained a high quality Gini coefficient for 43 countries. When we did not find a high quality Gini coefficient with this procedure, we have relied on the other (low quality) data available in this dataset following the same procedure. When more than one coefficient was available in the same year, we have taken the average. This has allowed us to obtain a Gini coefficient for a total of 57 countries.³⁶ The data from Deininger and Squire specify if the Gini coefficient is computed using information on income or on expenditure, if the income is gross or net of taxes and if the recipient unit is an individual or a household. Deininger and Squire argue that the most important distinction is between the Gini coefficients that are based on information on income and it but the parliament must approve it. See PT (2003, ch. 4) for details and clarifications.

³⁵The choice of taking a Gini coefficient that refers to few years later the constitutional choice is based on the same rationale that has led PT (2003) to exclude the reforms that some countries have made during the 1990s in their cross-sectional analysis for the period 1990-98. They argue that it takes some time before constitutional reforms have an impact on fiscal policy outcomes (see p. 88). However, if this is true (as we believe), then there will also need some time before the constitution has an impact on income inequality given that this effect should mainly work through fiscal policy.

³⁶Cyprus and South Africa are not in the dataset although we had a Gini coefficient for these countries. Cyprus has been excluded because we had a Gini coefficient for one year only and two constitutional choices very close to that year. While the more reasonable constitution would in principle favor our hypothesis, including Cyprus with either constitutions leaves our results unchanged. Details can be found in the data appendix. The exclusion of South Africa can be justified on both empirical and theoretical grounds. Empirically, it can be shown that it is an outlier and an influential observation. An additional appendix (available from the authors) shows that the inclusion of this observation in the sample has a very large impact on the estimates. This is not surprising given that South Africa is a parliamentary democracy with a proportional electoral system and has a very high Gini coefficient (62.3). South Africa is composed mainly by two groups, rich-white and poor-black, with the latter having a large absolute majority. This violates the assumption of our theory that there is not a well defined group which alone accounts for more than 50% of the population. Finally, notice that our sample does not contain the oldest democracies like U.S., U.K. or the Scandinavian and Northern European countries. However, we have provided historical evidence for these countries which appears to be consistent with our theory.

those based on expenditure. In order to ensure intertemporal and international comparability, they strongly suggest to make the adjustment for differences between income-based and expenditure-based coefficients by adding 6.6 points to the latter. We have therefore made this adjustment and denoted with *giniyai* the Gini coefficient with this correction.

An important point that we need to address concerns the classification of the countries into majoritarian and consensual democracies. As explained before, this classification is straightforward for parliamentary systems given that the executive is accountable to the parliament so that the electoral rule, majoritarian versus proportional, is enough to classify the type of democracy. In presidential regimes fiscal policy is the outcome of a bargaining process between the president and the legislative assembly. If the assembly is elected with plurality rule, then it is reasonable to classify these systems as majoritarian democracies given that both subjects that have a role in fiscal policy decisions are elected with a winner-takes-all process and are, therefore, expected to have the same fiscally conservative preferences. Instead, when the congress is elected with PR, it is important to understand the power of the president in fiscal policy decisions. We now argue that the presidential systems with PR in our dataset are characterized by a relative powerful president and should, therefore, be considered majoritarian democracies.³⁷

First, we remind that we take the classification of PT who classify as presidential only those regimes where the confidence of the assembly is not necessary for the executive and, therefore, consider most semi-presidential and premier-presidential systems as parliamentary democracies. Second, the presidential countries with PR in our dataset are classified as "direct presidential" in the Database of Political Institutions compiled by Beck et al. (2000). The three elements they consider are: a) veto power: president can veto legislation and the parliament needs a supermajority to override the veto; b) appoint prime minister: president can appoint and dismiss prime minister and/or other ministers; c) dissolve parliament: president can dissolve parliament and call for new elections. The system is a "direct presidential" if (a) is true, or if (b) and (c) are true. Third, notice that presidential countries with an assembly elected with PR are typically

³⁷In an extension of our model (available from the authors) we show that when the president has a significant power in the making of fiscal policy, a presidential regime with PR is essentially equivalent to our model of majoritarian democracy. Very briefly, we assume in this extension that the president acts as (first) agenda setter, offering a policy proposal to the legislature. If this proposal is not accepted by a parliamentary majority, the leader of the party having the largest number of seats in the legislature is nominated (second) agenda setter, and has a chance of forming a majority itself. If no agreement is reached within the second bargaining round, or if the president exercises its veto right on the policy approved by the parliament, a status quo policy is implemented. In equilibrium, the president is a rich and the policy implemented is its preferred policy, or a policy very close to that one, under many different specifications of the status quo.

observed in Latin America. 15 out of 17 countries with this system in our sample are located there.³⁸ While the exact distribution of the legislative power varies across countries, according to many scholars of comparative politics the president in Latin American countries typically plays a key role in the legislative process regarding fiscal policy. For instance, Morgenstern (2002) argues that Latin American legislatures are only "reactive", namely they only have the ability to amend/veto legislative proposals made by the president. Such prerogatives are also limited as presidents facing a hostile legislature typically have many ways to bypass it by using their "unilateral powers", such as various types of decrees and regulatory rule making, as well as their own veto powers on parliamentary deliberations.³⁹

Based on the above considerations, we have structured our empirical analysis as follows. First, we have analyzed the relationship between income inequality and electoral system in the sub-sample of parliamentary democracies. This choice is justified by the fact that our model describe very precisely this system and also allows us to consider a more homogenous sample. Second, we have considered the whole sample classifying the type of democracy on the base of the electoral system only. In this case, the presidential systems with PR enter as consensual democracies. As we will see, this is the worst specification for our theory because the income distribution in these countries is generally very unequal. Finally, we develop the empirical work with the classification that we consider more reasonable and closer to our model, namely by classifying the presidential regimes with PR as majoritarian systems. This implies that countries that are parliamentary with PR enter as consensual democracies while presidential regimes and parliamentary with plurality rule are all classified as majoritarian.

Table 1 provides descriptive statistics for income inequality in various samples. Among the parliamentary democracies, the countries with a majoritarian electoral rule are more unequal than those with PR. The difference in the average Gini coefficient is almost 10 points and it is statistically significant at 1% level. The unconditional correlation between Gini coefficient and majoritarian electoral system is 0.485. The table also reports the average Gini coefficient for presidential regimes when the assembly is elected with majoritarian

³⁸ These countries are Argentina, Bolivia, Brazil, Colombia, Dominican Republic, Ecuador, El Salvador, Guatemala, Honduras, Mexico, Nicaragua, Paraguay, Peru, Uruguay, Venezuela. The two exceptions are South Korea and Sri Lanka.

³⁹According to Cox and Morgenstern (2002, p. 461) "...presidents in Latin America regularly make policy decisions almost unilaterally. Presidents in Bolivia, Brazil, Chile, Colombia, Ecuador, Paraguay, Peru, Uruguay and Venezuela have tremendous advantages in structuring the budget process, as the legislatures there are constitutionally restricted from making significant changes." As a result, according to the same authors, "Latin American legislatures are hindered by a lack of time, resources and experience. This combination of constitutional and organizational limits has converted many Latin American presidents into virtual budget dictators."

and proportional electoral system. We observe that the average income inequality is high (the Gini index is about 49) and independent on the electoral formula of the assembly. In our view, this is consistent with the majoritarian characteristics of both systems. When we consider the whole sample and distinguish on the base of the electoral rule, we find that countries with plurality rule have higher income inequality than those with PR but this difference is lower with respect to the sample of parliamentary democracies and its significance is at 10% level. The unconditional correlation between the Gini coefficient and the electoral system is positive but not very high (i.e. it is 0.174). Finally, the last part of the table shows the statistics when presidential systems with PR are classified as majoritarian democracies. Majoritarian democracies are much more unequal than consensual ones and the difference in average inequality is about 13 points with a statistically significance at 1% level. As expected, there is also a high correlation (0.532) between Gini coefficient and majoritarian democracy.

Panel A of Table 2 reports the majoritarian and consensual countries in each quartile of the income distribution for the sample of parliamentary democracies. The first quartile contains just one majoritarian country and seven consensual ones, while we observe the opposite pattern in the last quartile of the distribution. The number of majoritarian (consensual) countries is also monotonically increasing (decreasing) in the degree of income inequality. Panel B shows the results of the same analysis for the whole sample when we consider the classification based of the electoral system only. As expected from the above discussion, we observe a positive relationship between degree of income inequality and number of majoritarian democracies, though this is weaker than in the sample of parliamentary democracies. The positive link between inequality and majoritarian electoral system can be appreciated if we consider two groups, countries with inequality below and above the median. The number of majoritarian countries with income inequality higher than the median is the double of those with inequality lower than the median. Finally, Panel C provides the results of this analysis when presidential regimes with an assembly elected with PR are classified as majoritarian systems. The positive relationship between income inequality and majoritarian democracy is very strong as, with the exception of one country, there are no consensual democracies with inequality above the median. We can therefore conclude that the unconditional relationship between income inequality and constitution is strongly supportive of our theory.

We now present the results of logit regressions when the relationship between income inequality and the type of constitution is analyzed conditioning for different variables that may potentially affect the choice of the constitution. The most important variable is ethnic fragmentation. Lijphart (1999) suggests the existence

of a positive relationship between PR and ethnic fragmentation. Instead, Aghion, Alesina and Trebbi (2004) find a negative correlation and reconcile the different results with the fact that Lijphart considers a smaller number of countries and democracies only. Hence, we always control for ethnic fragmentation and use the variables ethnic and language from Alesina et al. (2004). The advantage in using these two variables is related to the possibility of disentangling different kinds of heterogeneity as the first one is based on a broad measure of ethnicity while the second is strictly based on language. Alesina et al. make clear that these data come from the 1990s but there is no problem of endogeneity for a 30 years horizon, which approximately coincide with the period of our analysis.⁴⁰ We also control for religious fractionalization using the variable religion taken from the same authors.⁴¹ However, we do not include it in the baseline specification because Alesina et al. argue that the endogeneity problems for this variable may be more serious.

We control for country size and level of development by including the log of population in 1960 ($lpop_60$) and the GDP per capita expressed relative to the U.S. in 1960 (y_60) taken from the Penn World Table 6.1.⁴² Indeed, Lijphart (1977) argues that smaller countries are potentially exposed to more serious external threats by foreign powers and therefore more likely to promote intersegmental cooperation by adopting consensual constitutions. While we do not have in mind any specific causal mechanism from economic development to constitutional choice, we recognize that diverse traditions of thought in political sociology including modernization theory argue the existence of some link between income and political institutions.

We consider three regional dummies to take into account the role of geography. In particular, we use a dummy variable for continental location in Africa (africa), in eastern and southern Asia (asiae) and southern and central America including the Caribbean (laam). To take into account the possible effects of religious and cultural factors, we control for the percentage of the population in each country professing the Protestant religion in 1980 (prot80), the percentage of the population belonging to the Roman Catholic religion in 1980 (catho80), and if the majority of population is Confucian-Buddhist-Zen (confu). PT (2003, 2004) argue that

 $^{^{40}}$ Only few countries in our sample have a constitution chosen before 1960 and, in any case, all after WWII.

⁴¹The variables *ethnic*, *language* and *religion* take values in the range between zero and one that are increasing in the degree of ethnic, linguistic and religious fractionalization respectively. There is no data for the variable *language* for El Salvador.

⁴²There are a few countries for which the data on the level of income per capita relative to the U.S. and the size of population are not available in 1960. In these cases, we have taken the first year available. This is equivalent to assuming that the ratio between the GDP per capita of these countries and that one of the U.S. has remained unchanged between 1960 and the first year available. As a robustness check, we have also run regressions with the log of the GDP per capita in 1960. For the countries where GDP per capita in 1960 was not available, we have multiplied their per capita income relative to the U.S. in the first year available with U.S. per capita income in 1960. Using the log of this variable instead of the GDP per capita expressed relative to U.S. yields basically the same results.

ex-British colonies tend to be parliamentary and majoritarian while all former Spanish-Portuguese colonies are presidential. To take into account these effects, we control for British (col_uk) , for Spanish-Portuguese (col_esp) and other (col_oth) colonial origins. All the above variables are taken from the PT dataset. Aghion, Alesina and Trebbi (2004) use similar control variables in their empirical analysis.

Table 3 presents the relationship between constitution and income inequality in the sample of parliamentary democracies. Column (1) shows the results for our baseline specification where we consider the degree of income inequality and of ethnic fragmentation. Higher inequality is positively correlated with a majoritarian electoral system and this relationship is statistically significant at 1%. The quantitative effect of an increase in income inequality on the probability of adopting a majoritarian system is also very high as the marginal effect at the mean is 0.03.43 In columns (2) and (3) we control for the population and level of income respectively. In column (4), we include dummies for Africa and Asia while the dummy for Latin America is omitted because the four parliamentary democracies in this region are all majoritarian. Column (5) shows the results when we add the religious characteristics of each country that are likely to be correlated with cultural factors. In all specifications, we find that the relationship between inequality and constitution is unchanged with respect to the baseline one.⁴⁴ In column (6), we control for colonial origins and find that the correlation between income inequality and majoritarian democracy has the expected sign but it is not statistically significant at standard levels. This is due to the inclusion of the dummy variable for former British colonies.⁴⁵ In this sample, 8 out of 9 ex-British colonies have a majoritarian electoral rule. It is worth noting that these former British colonies are also very unequal. The mean Gini coefficient for these 9 countries is 46.58 and the most equal one has a coefficient of 38.1. This in turn implies that all former British colonies have a degree of inequality above average (see Table 1). Such collinearity between the Gini coefficient and British colonial origin, combined with measurement errors in inequality, could be at the root of inequality becoming insignificant. In other words, while controlling for colonial origins substantially weakens our results, it is not clear to us whether former colonies have chosen a certain type of democracy to imitate the colonizer's institutions or because these type of institutions suit well with their economic fundamentals. Finally, column (7) shows that the results are robust when we control for religious fractionalization.

Table 4 reports the results for the same analysis using the whole sample with the classification based on

⁴³We remind that the Gini coefficient employed is measured in a scale from 0 to 100. Therefore, this means that an increase of one point in the Gini coefficient increases the probability of adopting a majoritarian system by 3%.

 $^{^{44}}$ The Gini coefficient in column (5) is less precisely estimated but the p-value is 5.2% however.

⁴⁵There are no Spanish-Portuguese colonies in the sample and the inclusion of *col_oth* alone would not change the results with respect to the baseline specification. When we control only for *col_uk*, the Gini estimate is 0.02 and the p-value 12.4%.

the electoral rule. While the unconditional correlation between income inequality and electoral system in this sample is not very high, the correlation becomes strong when we control for ethnic fragmentation. In fact, in the baseline specification reported in column (1) we find that the marginal effect evaluated at the mean of an increase in inequality on the probability of adopting a majoritarian electoral rule is 0.016 and this is statistically significant at 5% level. The linguistic fractionalization coefficient has also a positive statistically significant effect. The same result is obtained controlling for population and income (columns (2) and (3)). The Gini coefficient has the same magnitude but it is less precisely estimated when we include the three regional dummies. The p-value of the coefficient is 10.7% however. In columns (5) and (6) we control for religious and for colonial origins respectively. The estimated Gini coefficient is pretty large (about 0.03) and significant at 5% in both specifications. Results are also robust when controlling for religious fractionalization which is positively correlated with plurality rule (see column (7)).

Table 5 shows the results when presidential regimes with PR are classified as majoritarian democracies. The relationship between income inequality and majoritarian democracy is very strong. The marginal effect on the probability of adopting a majoritarian democracy evaluated at the mean is large, about 0.024, and statistically significant at 1% level in all specifications.⁴⁶

Finally, we have performed the following three robustness checks. First, we have estimated the relationship between inequality and constitution also with probit and linear probability model. This yielded similar results. Second, we have used three other Gini coefficients: the unadjusted Gini; the Gini adjusted if the data is based on net income rather than gross (along with the adjustment for income versus expenditure); the Gini coefficient with a further adjustment if the information is based on households instead than individuals (see the data appendix for details). Using these Gini coefficients yields very similar estimates in all samples. Third, we have tried to understand if the quality of the Gini coefficients could play a role in our estimate even though, in principle, the inclusion of lower quality data should work against our hypothesis. To this purpose, we have first restricted the sample to high quality data only and the estimates are similar to those presented above. A second check was to add a dummy variable for the quality (high or low) of the Gini coefficient in each specification. Again, the estimate for income inequality remained basically unchanged while the estimate of the variable for the quality of the Gini coefficient was always statistically insignificant. The results of the above robustness checks are available from the authors.

⁴⁶We have also replicated the same analysis excluding the presidential regimes with PR. The results (available from the authors) are very close to those obtained for the sample of parliamentary democracies reported in Table 3.

Therefore, we conclude that there is a strong positive relationship between income inequality and majoritarian democracy as predicted by our model.

9 Constitutions and Politics in Contemporary Democracies

In this Section, we discuss the available empirical evidence confirming other results of our model. A first prediction of our theory is that consensual democracies should have bigger governments than majoritarian ones and a larger part of government expenditure should go to the advantage of a greater number of social groups and, in particular, to lower income individuals (the poor and the middle class). These results are in line with the recent theoretical literature, e.g. Persson and Tabellini (2000, ch. 8 and 9), Lizzeri and Persico (2001), Milesi-Ferretti, Perotti and Rostagno (2002), that generally predicts that proportional electoral systems should be associated with the provision of more public goods, larger and more universalistic welfare programs, and a larger overall size of government, arguably preferred by the poor and by the middle class. The empirical literature on this topic also confirms our results. Persson and Tabellini (2003, 2004) present cross-country evidence suggesting that a switch from proportional to majoritarian elections reduces total government spending by almost 5% of GDP and welfare spending by 2-3% of GDP. Results are similar for a switch from a parliamentary to a presidential regime. Milesi-Ferretti, Perotti and Rostagno (2002) and Lijphart (1999) provide other empirical evidence going in the same direction.

Our contribution can also shed some light on the argument usually made that public expenditure is higher under proportional electoral systems than under majoritarian ones because the former favor the representation of many groups and the formation of multi-party coalition governments, which in turn spend more because they need to please broader and more diverse constituencies than single-party executives. Our model has the same prediction, but suggests that the greater level of public expenditure observed in consensual democracies is due to the *selection bias* in the composition of the government coalition. Consensual democracies should be expected to be ruled relatively more often by center-left coalitions, while the fiscally conservative right should have an advantage in majoritarian constitutions. A large body of empirical evidence both in political science and economics about the effects of the government's ideology on fiscal policy outcomes suggests that left-wing executives are willing to tax and spend more than right-wing ones (see on this the comprehensive study of Alesina, Roubini and Cohen, 1997, and the references therein cited, as well as the recent work of Perotti and Kontopoulos, 2002). Moreover, there is also some evidence that center-left

government coalitions are observed more often in PR systems and right-wing governments in majoritarian democracies (see Powell, 2002, and Ticchi and Vindigni, 2003).

The results of our model can also help us to better understand the relationship between the distribution of income and fiscal policy outcomes. Much of the research on this topic is based on the seminal contribution of Meltzer and Richard (1981), which has the well known reduced form prediction that a higher (pretax) inequality should be expected to generate the political support for a larger fiscal redistribution of resources. This theory, however, does not seem to be corroborated by the data (e.g. Perotti, 1996, Bénabou, 1996, Alesina, Glaeser and Sacerdote, 2001) which, if anything, indicate that more equal societies tend to redistribute more rather then less. Our theory suggests that the absence or a negative relationship between inequality and redistribution may be due to the fact that more unequal societies are more likely to choose a majoritarian system, which in turn generate lower redistribution. Moreover, income inequality may have ambiguous effects also for a given constitution. In a consensual democracy, an increase in inequality generally leads to an increase in taxation and redistribution for a given government coalition, but has the opposite effect if it leads to the formation of a center-right majority instead than a center-left one. In a majoritarian democracy, more inequality affects taxation and redistribution only if it is associated to a variation in the income of the rich.

10 Conclusions

This paper shares with other recent contributions in political economics the premise that constitutional principles are of great importance in shaping fiscal policy outcomes in representative democracies. We show that generally consensual democracies have higher taxation and size of government than majoritarian ones. In our model, this is driven by the *selection bias* in the composition of the government coalition as consensual democracies should be expected to be ruled more often by center-left coalitions, while the right should have an advantage under majoritarian constitutions. But on top of this we demonstrate that, once institutions are viewed as endogenous, consensual democracy is more difficult to sustain politically in a more unequal society since greater inequality tends to undermine the stability of the coalition supporting it.

The results of our model concerning the effects of inequality on institutional choice, as well as on fiscal policy, appear to be not only significant *per se*, but for the guidelines they offer for future empirical research on the economic consequences of constitutions. In particular, they provide a firm theoretical background for

the intuitive claim that the assignment of constitutions across countries is not random, but reflects a number of "fundamentals", among which the extent of income inequality is of special importance. The results of our empirical analysis strongly support this theoretical finding.

11 Appendix

11.1 Proof of Proposition 1

We first prove that only the rich run for office and then that the set of citizen-candidates running for office is not empty.

Step 1. Assume that at least one rich candidate runs for office. Would anyone else run for office? The answer is no. If a middle class agent also run, he would be defeated by the rich candidate because both the rich and the poor would vote for the rich. The poor find it convenient to vote for the rich candidate because in their eyes he is the less bad of the two of them: he offers to the poor none of their preferred public good but demands lower taxes. Similarly, if a poor agent runs for office against a rich candidate, he would be defeated by the vote of the middle class and the rich. Lastly, notice it cannot be the case that (at least) a candidate from each group runs for office. Indeed, the middle class candidate would win the election with certainty in that instance (recall that $m^b > \max\{m^p, m^r\}$ by assumption), and therefore neither a poor nor a rich candidate would run against him.

Step 2. We now demonstrate that the set of citizen-candidates running for office is not empty, i.e. that at least one rich candidate runs for office. Let p denote the probability of victory for a rich citizen-candidate (in a symmetric equilibrium, this will be the same across identical citizen-candidates). A rich agent wants to run for office if and only if the expected gain of running exceeds its cost, namely if $\{p[w^r(\tau^r, G^r) + k] + (1-p)w^r(\tau^r, G^r)\} - w^r(\overline{\tau}, \overline{G}) \geq \varepsilon$, where $(\overline{\tau}, \overline{G})$ denotes the policy vector implemented if he does not run and the term in the graph parenthesis is the expected utility if he runs, given that only rich candidates do so. To show that the set of citizen-candidates running for office is not empty, it is sufficient to demonstrate that this condition is satisfied when only one rich person runs for office \forall $(\overline{\tau}, \overline{G}) \in ([0,1] \times \Re_+^3)$. To see this, observe that in this case p=1 and the participation constraint of the rich candidate reads $w^r(\tau^r, G^r) + k - w^r(\overline{\tau}, \overline{G}) \geq \varepsilon$. Since the policy vector (τ^r, G^r) maximizes the welfare of the rich, $w^r(\tau^r, G^r) > w^r(\overline{\tau}, \overline{G})$, \forall $(\overline{\tau}, \overline{G}) \in ([0,1] \times \Re_+^3)$ with $(\overline{\tau}, \overline{G}) \neq (\tau^r, G^r)$, and given that $k \geq \varepsilon$, the participation constraint of the rich candidate always holds with strict inequality when p=1.

11.2 Proof of Proposition 2

In this Appendix we show that exist a value of ϕ , that we call ϕ^* , such that the utility derived by the middle class from the government coalition with the poor is higher than the corresponding utility from the coalition with the rich if $\phi < \phi^*$, and that the opposite is true whenever $\phi > \phi^*$. To this aim, we first show that the utility of the middle class in the government coalition with the rich $w^b_{1,b,r}$ is monotonically increasing in ϕ , while the utility of the middle class in the coalition with the poor $w^b_{1,b,p}$ is strictly monotonically decreasing in ϕ . Then, to prove that there is a single crossing between these two schedules in the range where $\phi \in (0,1)$, we show that $w^b_{1,b,p}$ is higher than $w^b_{1,b,r}$ at $\phi = 0$, and that the opposite holds at $\phi = 1$.

The utility of the middle class in the government coalition with the rich $w_{1,b,r}^b$ is defined by the maximization problem (5) subject to the participation constraint of the rich (4). Differentiating (5) with respect to ϕ and applying the envelope theorem we get that

$$\frac{dw_{1,b,r}^b}{d\phi} = -H_g(g_{1,b,r}^b) \frac{\partial g_{1,b,r}^r}{\partial \phi} \geqslant 0. \tag{11}$$

Indeed, assuming that (4) is binding and applying the implicit function theorem we get

$$\frac{\partial g_{1,b,r}^r}{\partial \phi} = -\frac{\left[(1 - \tau_{2,r,p}) y^r + H(g_{2,r,p}^r) \right] - (1 - \tau_{2,p,b}) y^r}{H_q(g_{1,b,r}^r)} < 0 \tag{12}$$

given that the numerator is positive because it is the difference between the utility of the rich when they are agenda setter at round 2 (which is greater than y^r) and their utility under the poor and middle class government coalition (which is lower than y^r). In this case $\frac{dw^b_{1,b,r}}{d\phi}$ is strictly positive. Instead, if the participation constraint of the rich (4) is not binding, then $w^b_{1,b,r}$ is at its global maximum, $\frac{\partial g^r_{1,b,r}}{\partial \phi} = 0$ and $\frac{dw^b_{1,b,r}}{d\phi} = 0$. Hence, the result in (11) shows that $w^b_{1,b,r}$ is monotonically increasing in ϕ .

The utility of the middle class in the government coalition with the poor $w_{1,b,p}^b$ is defined by the maximization problem (8) subject to the participation constraint of the poor (7). If we differentiate (8) with respect to ϕ and apply the envelope theorem we obtain that

$$\frac{dw_{1,b,p}^{b}}{d\phi} = -H_g(g_{1,b,p}^{b}) \frac{\partial g_{1,b,p}^{p}}{\partial \phi} < 0$$
 (13)

given that from (7)

$$\frac{\partial g_{1,b,p}^p}{\partial \phi} = -\frac{y^p - \left[(1 - \tau_{2,p,b}) y^p + H(g_{2,p,b}^p) \right]}{H_g(g_{1,b,p}^p)} > 0.$$
 (14)

Indeed, the numerator at the right hand side of (14) is negative because it is the difference between y^p and the utility of the poor when they are agenda setter at round 2 (which is greater than y^p). The result in (13) means that $w_{1,b,p}^b$ is strictly monotonically decreasing in ϕ .

Then, it remains to show that $w_{1,b,p}^b(0) > w_{1,b,r}^b(0)$ and that $w_{1,b,p}^b(1) < w_{1,b,r}^b(1)$. We prove the first inequality by showing that the participation constraints of the rich and the poor at $\phi = 0$ imply that, for any given level of tax rate chosen by the middle class, the rich have to be compensated with a greater amount of their preferred public good. Indeed, the participation constraint of the poor at $\phi = 0$ implies that

$$H(g_{1,b,p}^p) = \tau_{1,b,p} y^p \tag{15}$$

while from the participation constraint of the rich we get⁴⁷

$$H(g_{1,b,r}^r) = \left[(1 - \tau_{2,r,p}) y^r + H(g_{2,r,p}^r) \right] - (1 - \tau_{1,b,r}) y^r.$$
(16)

The expression in the square brackets in the right hand side of (16) is the utility of the rich when they are agenda setter at round 2 and it is clearly greater than y^r . Hence, it is of course the case that

$$H(g_{1\,h\,r}^r) > \tau_{1,b,r}y^r.$$
 (17)

By combining (15) and (17) and taking into account also the fact that $y^r > y^p$, one can easily verify that for any given level of tax rate $\tau_{1,b,r} = \tau_{1,b,p} = \tau$ we have that $g_{1,b,r}^r > g_{1,b,p}^p$. This in turn implies that the middle class agenda setter obtains a higher level of utility by making a government coalition with the poor instead than with the rich, i.e. $w_{1,b,r}^b(0) > w_{1,b,r}^b(0)$.

At $\phi = 1$ the utility of the middle class under the government coalition with the poor $w_{1,b,p}^b$ is equal to y^b . This result can be obtained by observing that at $\phi = 1$ the maximization problem of the middle class is subject to the participation constraint of the poor where they are agenda setter with probability one at round 2 and maximize their utility subject to the constraint of giving to the middle class a level of utility equal to the status quo (which corresponds to their level of income y^b).⁴⁸

$$\max_{\left\{\tau_{1,b,p}\right\}} w_{1,b,p}^b = (1 - \tau_{1,b,p}) y^b + H(\tau_{1,b,p} \overline{y} - g_{1,b,p}^p)$$

⁴⁷The participation constraint of the rich is always binding at $\phi = 0$ as their expected utility at round 2 is greater than y^r .

⁴⁸ Formally, the maximization problem when $\phi = 1$ is the following:

From the maximization problem of the middle class when she forms a government coalition with rich, it is immediate to verify that $w_{1,b,r}^b$ is always greater than y^b at $\phi=1$. Indeed, from the participation constraint of the rich at $\phi=1$, i.e. $(1-\tau_{1,b,r})\,y^r+H(g_{1,b,r}^r)\geqslant (1-\tau_{2,p,b})\,y^r$, we know that the middle class could implement the following policy: $0<\tau_{1,b,r}\leqslant\tau_{2,p,b},\,g_{1,b,r}^r=0,\,g_{1,b,r}^b=\tau_{1,b,r}\overline{y}$. This policy satisfies the participation constraint of the rich and gives to the middle class a higher utility than her income level y^b . Given that the optimal policy gives to the middle class a higher utility than this policy, and therefore of y^b , it is clear that $w_{1,b,r}^b$ (1) $> w_{1,b,p}^b$ (1).

11.3 Proof of Result 1

In this Appendix we show that under the assumptions stated in the main text the following inequalities hold for all ϕ : $\tau^r < \tau_{1,b,r} < \tau_{1,b,p}$. To this aim, we first prove that $\tau^r < \tau_{1,b,r} \leqslant \tau^b$ by showing that $\tau_{1,b,r} = \tau^r$ at $\phi = 0$, $\tau_{1,b,r} = \tau^b$ at $\phi = 1$, and that $\tau_{1,b,r}$ is monotonically increasing in ϕ . Next, we show that $\tau^b < \tau_{1,b,p}$. At $\phi = 0$ the rich are agenda setter with probability one at round 2, and they form the government coalition with the poor. If the income of the poor is equal to zero, the fiscal policy implemented at the second round corresponds to the dictatorial policy of the rich as the poor do not need to be compensated with a positive amount of their preferred public good. Thus, to form a government coalition at round 1, the middle class has to propose a policy to the rich such that their level of utility is the same they obtain at round 2. Given that the latter is equal to its global unconstrained maximum, the middle class can only implement the dictatorial policy of the rich, that is: $\tau_{1,b,r} = \tau_{2,r,p} = \tau^r$, $g^r_{1,b,r} = g^r_{2,r,p} = \tau^r \overline{y}$ and $g^b_{1,b,r} = 0$. In other words, the policy implemented (and the tax rate chosen) by the government coalition formed by the middle class and the rich at $\phi = 0$ (if $y^p = 0$) is the same as the one in majoritarian democracy.

We have previously shown that $\tau_{1,b,r} = \tau^b$ for all values of ϕ such that the participation constraint of the rich is not binding and that this is always the case at $\phi = 1$ if $y^p = 0$.

To prove that $\tau_{1,b,r}$ is monotonically increasing in ϕ , we differentiate the first order condition (6) with respect to ϕ by taking into consideration the fact that from the participation constraint of the rich (4),

$$s.t. \quad \left(1 - \tau_{1,b,p}\right)y^p + H(g^p_{1,b,p}) \geqslant \max_{\left\{\tau_{2,p,b}\right\}} \left(1 - \tau_{2,p,b}\right)y^p + H(\tau_{2,p,b}\overline{y} - g^b_{2,p,b})$$

s.t.
$$w_{2,p,b}^b \equiv (1 - \tau_{2,p,b}) y^b + H(g_{2,p,b}^b) = y^b$$

and it is immediate to verify that $w_{1,b,p}^b = w_{2,p,b}^b = y^b$.

 $g^r_{1,b,r} = g^r_{1,b,r} \left(\phi; \tau_{1,b,r}(\phi) \right)$ and $\frac{\partial g^r_{1,b,r}}{\partial \tau_{1,b,r}} = \frac{y^r}{H_g(g^r_{1,b,r})}$.⁴⁹ After some algebra we obtain

$$\frac{\partial \tau_{1,b,r}}{\partial \phi} \left\{ H_{gg}(g_{1,b,r}^b) \left[\overline{y} - \frac{y^r}{H_g(g_{1,b,r}^r)} \right]^2 + H_{gg}(g_{1,b,r}^r) \frac{(y^r)^2 H_g(g_{1,b,r}^b)}{\left[H_g(g_{1,b,r}^r) \right]^3} \right\} + \frac{\partial g_{1,b,r}^r}{\partial \phi} \left\{ -H_{gg}(g_{1,b,r}^b) \left[\overline{y} - \frac{y^r}{H_g(g_{1,b,r}^r)} \right] + H_{gg}(g_{1,b,r}^r) \frac{y^r H_g(g_{1,b,r}^b)}{\left[H_g(g_{1,b,r}^r) \right]^2} \right\} = 0.$$
(18)

which allows us to determine the sign of $\frac{\partial \tau_{1,b,r}}{\partial \phi}$. Indeed, notice that the term in graph parenthesis that multiplies $\frac{\partial \tau_{1,b,r}}{\partial \phi}$ is always negative and from (12) we know that $\frac{\partial g_{1,b,r}^r}{\partial \phi} < 0$. Therefore, the sign of the relationship between $\tau_{1,b,r}$ and ϕ depends on the sign of the term in the second graph parenthesis of (18) which multiplies $\frac{\partial g_{1,b,r}^r}{\partial \phi}$. In particular, $\frac{\partial \tau_{1,b,r}}{\partial \phi} \geqslant 0$ if and only if

$$-H_{gg}(g_{1,b,r}^b)\left[\overline{y} - \frac{y^r}{H_g(g_{1,b,r}^r)}\right] + H_{gg}(g_{1,b,r}^r)\frac{y^r H_g(g_{1,b,r}^b)}{\left[H_g(g_{1,b,r}^r)\right]^2} \leqslant 0$$
(19)

and vice versa. Even though in general it is not possible to give a definite sign to the left hand side of (19), assuming a power function for the utility of the public good it turns out that inequality (19) is satisfied if

$$\frac{A(g_{1,b,r}^r)^{\alpha}}{\tau_{1,b,r}y^r} \leqslant \frac{1}{\alpha} \tag{20}$$

where the numerator of the left hand side of (20) is the utility that the rich get from their preferred public good provided and the denominator represents the taxes they pay. It is useful to rewrite inequality (20) as

$$A(g_{1,b,r}^r)^{\alpha} - \tau_{1,b,r} y^r \leqslant (1 - \alpha) A(g_{1,b,r}^r)^{\alpha}. \tag{21}$$

By using a power function specification for $H(\cdot)$ and substracting y^r to both sides, we can rewrite the participation constraint of the rich (4) as

$$-\tau_{1,b,r}y^r + A(g_{1,b,r}^r)^\alpha = (1-\phi)\left[-\tau_{2,r,p}y^r + A(g_{2,r,p}^r)^\alpha\right] - \phi\tau_{2,p,b}y^r.$$
 (22)

The substitution of (22) into (21) leads to the following weak inequality

$$(1 - \phi) \left[-\tau_{2,r,p} y^r + A(g_{2,r,p}^r)^{\alpha} \right] - \phi \tau_{2,p,b} y^r - (1 - \alpha) A(g_{1,b,r}^r)^{\alpha} \leqslant 0.$$
 (23)

⁴⁹We remind that $\tau_{1,b,r} = \tau^b$ when the participation constraint is not binding.

Now, if we take into consideration the fact that the dictatorial policy of the rich is implemented at round 2 under the rich and poor coalition (see the discussion above), we are able to rewrite (23) as⁵⁰

$$(1-\phi)(1-\alpha)\alpha^{\frac{\alpha}{1-\alpha}}A^{\frac{1}{1-\alpha}}\left(\frac{\overline{y}}{y^r}\right)^{\frac{\alpha}{1-\alpha}} - \phi\tau_{2,p,b}y^r - (1-\alpha)A(g_{1,b,r}^r)^{\alpha} \leqslant 0.$$
 (24)

Notice that only the first component of the left hand side of (24) is positive and this term decreases until zero as y^r increases.⁵¹ Therefore, there exists a y^r sufficiently high relatively to \overline{y} such that (24) is always satisfied, which in turn means that $\frac{\partial \tau_{1,b,r}}{\partial \phi} \geqslant 0.52$

At this point we know that $\tau_{1,b,r}$ is monotonically increasing in ϕ , that $\tau_{1,b,r}(0) = \tau^r$ and that $\tau_{1,b,r}(1) = \tau^r$ τ^b and this implies that $\tau^r < \tau_{1,b,r} \leqslant \tau^b$.

We now want to prove that $\tau_{1,b,p}$ is always higher than τ^b and to this aim we show that $\tau_{1,b,p} = \tau^b$ at $\phi = 0$ and that $\tau_{1,b,p}$ is increasing in ϕ . The first point is easily shown by observing that the middle class agenda setter in the coalition with the poor can implement her dictatorial policy at $\phi = 0$ if $y^p = 0$ because the participation constraint of the poor (7) is not binding. To prove the second point we differentiate the first order condition (9) with respect to ϕ by taking into account that from the participation constraint of the poor (7), $g_{1,b,p}^p = g_{1,b,p}^p\left(\phi; \tau_{1,b,p}(\phi)\right)$ and $\frac{\partial g_{1,b,p}^p}{\partial \tau_{1,b,p}} = \frac{y^p}{H_g(g_{1,b,p}^p)}$. After rearranging terms we get

$$\frac{\partial \tau_{1,b,p}}{\partial \phi} \left\{ H_{gg}(g_{1,b,p}^b) \left[\overline{y} - \frac{y^p}{H_g\left(g_{1,b,p}^p\right)} \right] + H_{gg}(g_{1,b,p}^p) \frac{\left(y^p\right)^2 H_g(g_{1,b,p}^b)}{\left[H_g\left(g_{1,b,p}^p\right)\right]^3} \right\} + \\$$

$$+\frac{\partial g_{1,b,p}^{p}}{\partial \phi} \left\{ -H_{gg}(g_{1,b,p}^{b}) \left[\overline{y} - \frac{y^{p}}{H_{g}(g_{1,b,p}^{p})} \right] + H_{gg}(g_{1,b,p}^{p}) \frac{y^{p} H_{g}(g_{1,b,p}^{b})}{\left[H_{g}(g_{1,b,p}^{p}) \right]^{2}} \right\} = 0.$$
 (25)

From (14) we know that $\frac{\partial g_{1,b,p}^p}{\partial \phi} > 0$ and, given that the term in the first graph parenthesis is always negative, the sign of $\frac{\partial \tau_{1,b,p}}{\partial \phi}$ depends on the sign of the term in the second graph parenthesis that multiplies $\frac{\partial g_{1,b,p}^p}{\partial \phi}$. If

$$-H_{gg}(g_{1,b,p}^b)\left[\overline{y} - \frac{y^p}{H_g(g_{1,b,p}^p)}\right] + H_{gg}(g_{1,b,p}^p)\frac{y^p H_g(g_{1,b,p}^b)}{[H_g(g_{1,b,p}^p)]^2} > 0$$
(26)

⁵⁰We are using the fact that $\tau_{2,r,p} = \tau^r$, $g_{2,r,p}^r = \tau^r \overline{y}$ with $\tau^r = \left(\frac{\alpha A \overline{y}^{\alpha}}{y^r}\right)^{\frac{1}{1-\alpha}}$.

⁵¹The first component of (23) and (24) represents the net gain (with respect to the status quo) of the rich in utility terms when they are agenda setter at round 2 multiplied by $(1-\phi)$. Clearly, this component goes to zero as ϕ tends to one.

⁵² Using the fact that the dictatorial policy of the rich is implemented at $\phi = 0$ and therefore $g_{1,b,r}^r(0) = \tau^r \overline{y}$, it is immediate to verify that the left hand side of (24) is equal to zero at $\phi = 0$. Moreover, notice that the left hand side of (24) is more likely to be positive when ϕ is small. In this case $\tau_{1,b,r}$ would be decreasing in ϕ . When y^r in not sufficiently high, the numerical simulations confirm that $\tau_{1,b,r}$ is decreasing for values of ϕ low and then it becomes increasing when ϕ is big enough.

then $\frac{\partial \tau_{1,b,p}}{\partial \phi} > 0$, and vice versa. In this case we don't need to use a power function specification for $H(\cdot)$ because using the fact that $y^p = 0$ and $g^p_{1,b,p} > 0$ for all $\phi > 0$, inequality (26) becomes $-H_{gg}(g^b_{1,b,p})\overline{y} > 0$, which is always satisfied given that $H_{gg}(\cdot) < 0$. This implies that $\tau_{1,b,p}$ is monotonically increasing in ϕ and its minimum level is equal to τ^b at $\phi = 0$.

References

- [1] Acemoglu, Daron and Robinson, James A. *Economic Origins of Dictatorship and Democracy*. Cambridge: Cambridge University Press, 2005, forthcoming.
- [2] Acemoglu, Daron and Robinson, James A. "Why Did the West Extend the Franchise? Democracy, Inequality, and Growth in Historical Perspective." Quarterly Journal of Economics, November 2000, 115(4), pp. 1167-99.
- [3] Aghion, Philippe and Bolton, Patrick. "Incomplete Social Contracts." *Journal of the European Economic Association*, March 2003, 1(1), pp. 38-67.
- [4] Aghion, Philippe; Alesina, Alberto and Trebbi, Francesco. "Endogenous Political Institutions." Quarterly Journal of Economics, May 2004, 119(2), pp. 565-611.
- [5] Alesina, Alberto; Devleeschauwer, Arnaud; Easterly, William; Kurlat, Sergio and Wacziarg, Romain. "Fractionalization." Journal of Economic Growth, 2003, 8, pp. 155-194.
- [6] Alesina, Alberto and Glaeser, Edward. Fighting Poverty in the US and Europe: A World of Difference. Oxford, Oxford University Press, 2004.
- [7] Alesina, Alberto; Glaeser, Edward and Sacerdote, Bruce. "Why Doesn't the United States Have a European-Style Welfare State?" *Brookings Papers on Economic Activity*, 2001, 2.
- [8] Alesina, Alberto and Rosenthal, Howard. "A Theory of Divided Government." *Econometrica*, November 1996, 64(6), pp. 1311-42.
- [9] Alesina, Alberto and Rosenthal, Howard. Partisan Politics, Divided Government and the Economy. Cambridge, Cambridge University Press, 1995.
- [10] Alesina, Alberto; Roubini Nouriel and Cohen, Gerald D. Political Cycles and the Macroeconomy. Cambridge, MA: The MIT Press, 1997.

- [11] Austen-Smith, David. "Redistributing Income Under Proportional Representation." Journal of Political Economy, December 2000, 108(6), pp. 1235-69.
- [12] Austen-Smith, David and Banks, Jeffrey S. "Elections, Coalitions, and Legislative Outcomes." *American Political Science Review*, June 1988, 82, pp. 405-22.
- [13] Baldez, Lisa and Carey, John M. "Executive Agenda Control and Spending Policy: Lessons from General Pinochet's Constitution." American Political Science Review, January 1999, 43, pp. 29-55.
- [14] Barbera, Salvador and Jackson, Matthew O. "Choosing How to Choose: Self-Stable Majority Rules and Constitutions." *Quarterly Journal of Economics*, August 2004, 119(3), pp. 1011-1048.
- [15] Beard, Charles A. An Economic Interpretation of the Constitution of the United States. New York: The Macmillan Company, 1913.
- [16] Beck, Thorsten; Clark, George; Groff, Alberto; Keefer, Philip and Walsh, Patrick. "New Tools and New Tests in Comparative Political Economy: The Database of Political Institutions." World Bank Research WP No. 2283, 2000.
- [17] Bénabou, Roland. "Inequality and Growth." In Ben S. Bernanke and Julio J. Rotemberg (eds.), National Bureau of Economic Research Macroeconomics Annual, 1996, vol. 11, pp. 11-74. Cambridge: MIT Press.
- [18] Besley, Timothy and Coate, Stephen. "An Economic Model of Representative Democracy." Quarterly Journal of Economics, February 1997, 112(1), pp. 85-114.
- [19] Besley, Timothy and Coate, Stephen. "Public Provision of Private Goods and The Redistribution of Income." American Economic Review, September 1991, 81(4), pp. 979-84.
- [20] Cox, Gary W. and Morgenstern, Scott. "Epilogue: Latin America's Reactive Assemblies and Proactive Presidents," in Scott Morgenstern and Benito Nacif, eds, Legislative Politics in Latin America, Ch. 15. Cambdrige: Cambridge University Press, 2002.
- [21] Cusack, Thomas. "Partisan Politics and Public Finance: Changes in Public Spending in the Industrialized Democracies, 1955-1989." *Public Choice*, 1997, 91(3/4). pp. 375-395.
- [22] Deininger, Klaus and Squire, Lyn. "A New Data Set Measuring Income Inequality." World Bank Economic Review, 1996, 10, pp. 565-591.

- [23] Fernandez, Raquel and Rogerson, Richard. "On the Political Economy of Education Subsidies." *Review of Economic Studies*, April 1995, 62(2), pp. 249-62.
- [24] Lakeman, Enid and James, Lambert. Voting in Democracies. A Study of Majority and Proportional Systems. London: Faber and Faber, 1969.
- [25] Lijphart, Arend. Democracy in Plural Societies. A Comparative Exploration. New Haven: Yale University Press, 1977.
- [26] Lijphart, Arend. Patterns of Democracy. New Haven: Yale University Press, 1999.
- [27] Linz, Juan J. "Presidential or Parliamentary Democracy: Does It Make a Difference?" in Linz, Juan J. and Valenzuela, Arturo, eds, The Failure of Presidential Democracy, Baltimore: The Johns Hopkins University Press, 1994, pp. 3-87.
- [28] Lizzeri, Alessandro and Persico, Nicola. "The Provision of Public Goods Under Alternative Political Regimes." American Economic Review, March 2001, 91(1), pp. 225-39.
- [29] Meltzer, Allan H. and Richard, Scott F. "A Rational Theory of the Size of Government." Journal of Political Economy, October 1981, 89(5), pp. 914-27.
- [30] Messner, Matthias and Polborn Mattias K. "Voting on Majority Rules." Review of Economic Studies, January 2004, 71(1), pp. 115-132.
- [31] Milesi-Ferretti, Gian Maria; Perotti, Roberto and Rostagno, Massimo. "Electoral Systems and Public Spending." Quarterly Journal of Economics, May 2002, 117(2), pp. 609-657.
- [32] Morgenstern, Scott. "Explaining Legislative Politics in Latin America," in Scott Morgenstern and Benito Nacif, eds, *Legislative Politics in Latin America*, Ch. 14. Cambdrige: Cambridge University Press, 2002.
- [33] Morrisson, Christian, "Historical Perspectives on Income Distribution: the Case of Europe", in A.B. Atkinson and F. Bourguignon (eds.), Handbook of Income Distribution vol.1, Amsterdam: North-Holland, 2000.
- [34] Myerson, Roger B. "Incentives to Cultivate Favored Minorities under Alternative Electoral Systems."

 American Political Science Review, December 1993, 87(4), pp. 856-69.

- [35] Osborne, Martin J. and Slivinsky, Al. "A Model of Political Competition with Citizen-Candidates." Quarterly Journal of Economics, February 1996, 111(1), pp. 65-96.
- [36] Perotti, Roberto. "Growth, Income Distribution, and Democracy: What the Data Say." *Journal of Economic Growth*, June 1996, 1(2), pp. 149-87.
- [37] Perotti, Roberto and Kontopoulos, Yianos. "Fragmented Fiscal Policy." Journal of Public Economics, November 2002, 86(2), pp. 191-222.
- [38] Persson, Torsten; Roland Gérard and Tabellini Guido. "Separation of Powers and Political Accontability." Quarterly Journal of Economics, November 1997, 112(4), pp. 1163-1202.
- [39] Persson, Torsten; Roland Gérard and Tabellini Guido. "Comparative Politics of Public Finance." *Journal of Political Economy*, December 2000, 108(6), pp. 1121-61.
- [40] Persson, Torsten and Guido Tabellini. "Constitutional Rules and Fiscal Policy Outcomes." *American Economic Review*, March 2004, 94(1), pp. 25-45.
- [41] Persson, Torsten and Tabellini, Guido. Political Economics: Explaining Economic Policy. Cambridge, MA: The MIT Press, 2000.
- [42] Persson, Torsten and Tabellini, Guido. The Economic Effects of Constitutions. Cambridge, MA: The MIT Press, 2003.
- [43] Piketty, Thomas. "Income Inequality in France 1901-1998." Journal of Political Economy, 2003, 111(5), pp. 1004-42.
- [44] Powell, G. Bingham Jr. "PR, the Median Voter, and Economic Policy: An Exploration." Mimeo, University of Rochester, 2002.
- [45] Romer, Thomas and Rosenthal, Howard. "Bureaucrats Versus Voters: On the Political Economy of Resource Allocation by Direct Democracy." Quarterly Journal of Economics, 1979, 93(4), pp. 563-87.
- [46] Shugart, Matthew Soberg and Carey, John M. Presidents and Assemblies. Constitutional Design and Electoral Systems. Cambridge: Cambridge University Press, 1992.
- [47] Ticchi, Davide and Vindigni, Andrea. "Endogenous Constitutions." IIES Seminar Paper No. 726, Stockholm University, November 2003. http://www.iies.su.se/publications/seminarpapers/726.pdf

Table 1. Descriptive statistics of income inequality (variable: giniycai).

| Sample: PARLIAMENTARY DEMOCRACIES | | | | | | | | | |
|-----------------------------------|-------------|--------------------|------------------------|-------------------|----------|--|--|--|--|
| Electoral system | Obs | Mean | Std. Dev. | Min | Max | | | | |
| Majoritarian | 14 | 43.936 | 8.933 | 25.71 | 57.4 | | | | |
| Proportional | 17 | 34.179 | 9.132 | 21.5 | 51 | | | | |
| Total | 31 | 38.585 | 10.17 | 21.5 | 57.4 | | | | |
| Sample: PRESIDENTIAL DEMOCRACIES | | | | | | | | | |
| Electoral system | Obs | Mean | Std. Dev. | Min | Max | | | | |
| Majoritarian | 9 | 48.95 | 12.723 | 28.53 | 68.6 | | | | |
| Proportional | 17 | 49.534 | 9.33 | 33.64 | 65.38 | | | | |
| Total | 26 | 49.332 | 10.373 | 28.53 | 68.6 | | | | |
| | Sample: WI | HOLE – classifica | ntion: based on the el | ectoral system | | | | | |
| | Obs | Mean | Std. Dev. | Min | Max | | | | |
| Majoritarian | 23 | 45.898 | 10.596 | 25.71 | 68.6 | | | | |
| Consensual | 34 | 41.856 | 11.974 | 21.5 | 65.38 | | | | |
| Total | 57 | 43.487 | 11.515 | 21.5 | 68.6 | | | | |
| Sample: | WHOLE – cla | ssification: presi | dential systems with | PR enter as major | ritarian | | | | |
| | Obs | Mean | Std. Dev. | Min | Max | | | | |
| Majoritarian | 40 | 47.443 | 10.118 | 25.71 | 68.6 | | | | |
| Consensual | 17 | 34.179 | 9.132 | 21.5 | 51 | | | | |
| Total | 57 | 43.487 | 11.515 | 21.5 | 68.6 | | | | |

Table 2. Income distribution and constitution selection.

Panel A. Income distribution in parliamentary democracies.

| | 1 st quartile 21.5≤gini≤26.98 | 2 nd quartile 30.06≤gini≤40.58 | 3 rd quartile 41.71≤gini≤45.49 | 4 th quartile 46.02≤gini≤57.4 | Total |
|--------------|--|---|--|--|-------|
| Majoritarian | Ukraine | Nepal, Bangladesh, Malaysia India, Bahamas, Barbados | | Trinidad&T., Japan, Singapore, France, Thailand, Jamaica, Botswana | |
| Consensual | Slovak R., Hungary, Romania, Bulgaria, Czech R., Poland, Latvia | Taiwan, Spain, Italy, Germany, Portugal | Greece, Estonia, Fiji, Senegal | Turkey | |
| Majoritarian | 1 | 3 | 3 | 7 | 14 |
| Consensual | 7 | 5 | 4 | 1 | 17 |
| Total | 8 | 8 | 7 | 8 | 31 |
| | Below | the median | Above t | he median | |
| Majoritarian | | 4 | | 10 | 14 |
| Consensual | | 12 | 5 | | 17 |
| Total | | 16 | | 15 | 31 |

Panel B. Income distribution in the whole sample. Classification of majoritarian and consensual based on the electoral system.

| | 1 st quartile 21.5≤gini≤35.45 | 2 nd quartile 37.11≤gini≤43 | 3 rd quartile 45≤gini≤50.11 | 4 th quartile 51≤gini≤68.6 | Total |
|--------------|---|---|---|---|-------|
| Majoritarian | Ukraine, Belarus, Nepal, Bangladesh | Pakistan, Malaysia, Ghana, India | Bahamas, Barbados, Trinidad&T.,Philippines Japan, Uganda, Singapore, France, Zambia | Thailand, Jamaica, Botswana, Chile, Zimbabwe, Malawi | |
| Consensual | Slovak R., Hungary, Romania, Bulgaria, Czech R., Poland, Latvia, Taiwan, South Korea, Sri Lanka | Spain, Italy, Paraguay, Germany, Portugal, Uruguay, Greece, Argentina, Estonia, Fiji, Senegal | Dominican R,Venezuela El Salvador, Bolivia, Guatemala | Turkey, Peru, Brazil, Mexico, Nicaragua, Colombia, Honduras, Ecuador | |
| Majoritarian | 4 | 4 | 9 | 6 | 23 |
| Consensual | 10 | 11 | 5 | 8 | 34 |
| Total | 14 | 15 | 14 | 14 | 57 |
| | Below t | he median | Above th | e median | |
| Majoritarian | | 8 | | 5 | 23 |
| Consensual | | 21 | | 3 | 34 |
| Total | | 29 | 2 | 8 | 57 |

Panel C. Income distribution in the whole sample. Classification: presidential regimes with PR are classified as majoritarian democracies.

| | 1 st quartile 21.5≤gini≤35.45 | 2 nd quartile 37.11≤gini≤43 | 3 rd quartile 45≤gini≤50.11 | 4 th quartile 51≤gini≤68.6 | Total |
|--------------|--|--|---|--|-------|
| Majoritarian | Ukraine, Belarus, Nepal, South Korea, Sri Lanka, Bangladesh | Pakistan, Malaysia, Paraguay, Ghana, Uruguay, Argentina, India | Dominican R., Bahamas, Barbados, Trinidad&T., Venezuela, Philippines, Japan, Uganda, Singapore, El Salvador, France, Bolivia, Guatemala, Zambia | Thailand, Jamaica, Peru, Brazil, Mexico, Nicaragua, Botswana, Chile, Colombia, Honduras, Zimbabwe, Ecuador, Malawi | |
| Consensual | Slovak R., Hungary, Romania, Bulgaria, Czech R., Poland, Latvia, Taiwan | Spain, Italy, Germany, Portugal, Greece, Estonia, Fiji, Senegal | | Turkey | |
| Majoritarian | 6 | 7 | 14 | 13 | 40 |
| Consensual | 8 | 8 | 0 | 1 | 17 |
| Total | 14 | 15 | 14 | 14 | 57 |
| | Below t | he median | Above th | e median | |
| Majoritarian | | 13 | | 7 | 40 |
| Consensual | | 16 | 1 | [| 17 |
| Total | | 29 | 2 | 8 | 57 |

Table 3. Constitution selection and income inequality in parliamentary democracies.

| Sample: Parliamentary democracies – Logit estimates | | | | | | | | |
|---|--------------------|---------------------|-------------------|--------------------|-------------------|-------------------|---------------------|--|
| | (1) | (2) | (3) | (4) | (5) | (6) | (7) | |
| Dep. var. | maj | maj | maj | maj | maj | maj | maj | |
| giniycai | 0.03 (0.012)*** | 0.031 (0.012)*** | 0.03 (0.012)** | 0.033 (0.016)** | 0.032 (0.016)* | 0.02 (0.015) | 0.033 (0.012)*** | |
| ethnic | 0.16 (0.649) | 0.469 (0.911) | 0.152 (0.637) | 0.762 (0.715) | 0.339 (0.676) | -0.517 (0.818) | -0.129 (0.703) | |
| language | 0.233 (0.504) | 0.031 (0.634) | 0.096 (0.531) | -0.758 (0.759) | -0.2 (0.596) | 0.561 (0.711) | 0.486 (0.506) | |
| lpop_60 | | [0.626] | | | | | | |
| <i>y</i> _60 | | | [0.555] | | | | | |
| regional dummies | | | | [0.215] | | | | |
| prot80, catho80, confu | | | | | [0.234] | | | |
| colonial origins | | | | | | [0.238] | | |
| religion | | | | | | | [0.173] | |
| Observations Pseudo R-squared | 31 0.2 | 31 0.21 | 31 0.21 | 31 0.31 | 31 0.29 | 31 0.31 | 31 0.26 | |

Notes: Robust standard errors in parentheses. * Significant at 10%. ** Significant at 5%. *** Significant at 1%. Logit coefficients are marginal effects on the probability of having a majoritarian system evaluated at the mean. The table includes p-values for Chi-square test of joint significance of the control sets. Regional dummies include africa and asiae. Regional dummy laam has not been included because all four parliamentary democracies in Latin American are majoritarian. Colonial origins include col uk and col oth but not col esp as there are not Spanish-Portuguese colonies in the sample.

Table 4. Constitution selection and income inequality in the whole sample (classification based on the electoral rule).

| Whole sample – Logit estimates | | | | | | | | |
|----------------------------------|---------------------|---------------------|---------------------|------------------|---------------------|---------------------|---------------------|--|
| | (1) | (2) | (3) | (4) | (5) | (6) | (7) | |
| Dep. var. | maj | maj | maj | maj | maj | maj | maj | |
| giniycai | 0.016 (0.007)** | 0.016 (0.007)** | 0.017 (0.007)** | 0.015 (0.009) | 0.032 (0.014)** | 0.029 (0.013)** | 0.023 (0.009)*** | |
| ethnic | -0.755 (0.615) | -0.757 (0.615) | -0.729 (0.629) | -0.617 (0.57) | -0.81 (0.511) | -1.089 (0.525)** | -0.782 (0.504) | |
| language | 1.344 (0.498)*** | 1.344 (0.498)*** | 1.414 (0.499)*** | 0.843 (0.542) | 1.081 (0.402)*** | 1.277 (0.398)*** | 1.403 (0.411)*** | |
| lpop_60 | | [0.980] | | | | | | |
| y_60 | | | [0.665] | | | | | |
| regional dummies | | | | [0.518] | | | | |
| prot80, catho80, confu | | | | | [0.027] | | | |
| colonial origins | | | | | | [0.004] | | |
| religion | | | | | | | [0.012] | |
| Observations Pseudo R-squared | 56 0.21 | 56 0.21 | 56 0.21 | 56 0.24 | 56 0.39 | 56 0.43 | 56 0.34 | |

Notes: Robust standard errors in parentheses. * Significant at 10%. ** Significant at 5%. *** Significant at 1%. Logit coefficients are marginal effects on the probability of having a majoritarian system evaluated at the mean. The table includes *p*-values for Chi-square test of joint significance of the control sets. Regional dummies include *africa*, *asiae*, *laam*. Colonial origins include *col_uk*, *col_esp*, *col_oth*.

Table 5. Constitution selection and income inequality when presidential regimes with PR are classified as majoritarian democracies.

| Whole sample – Logit estimates | | | | | | | | |
|----------------------------------|---------------------|---------------------|---------------------|---------------------|---------------------|---------------------|---------------------|--|
| | (1) | (2) | (3) | (4) | (5) | (6) | (7) | |
| Dep. var. | maj | |
| giniycai | 0.024 (0.006)*** | 0.024 (0.006)*** | 0.023 (0.006)*** | 0.025 (0.008)*** | 0.025 (0.007)*** | 0.022 (0.008)*** | 0.025 (0.007)*** | |
| ethnic | 0.153 (0.283) | 0.222 (0.299) | 0.125 (0.267) | 0.339 (0.329) | 0.31 (0.311) | 0.06 (0.283) | 0.12 (0.296) | |
| language | 0.162 (0.199) | 0.129 (0.227) | 0.028 (0.222) | -0.071 (0.358) | 0.072 (0.214) | 0.054 (0.229) | 0.137 (0.208) | |
| lpop_60 | | [0.538] | | | | | | |
| y_60 | | | [0.167] | | | | | |
| regional dummies | | | | [0.297] | | | | |
| prot80, catho80, confu | | | | | [0.198] | | | |
| colonial origins | | | | | | [0.401] | | |
| religion | | | | | | | [0.261] | |
| Observations Pseudo R-squared | 56 0.28 | 56 0.29 | 56 0.30 | 56 0.33 | 56 0.34 | 56 0.32 | 56 0.31 | |

Notes: Robust standard errors in parentheses. * Significant at 10%. ** Significant at 5%. *** Significant at 1%. Logit coefficients are marginal effects on the probability of having a majoritarian system evaluated at the mean. The table includes p-values for Chi-square test of joint significance of the control sets. Regional dummies include africa and asiae. Regional dummy laam has not been included because all nineteen countries in Latin American enter as majoritarian. Colonial origins include col_uk and col_esp but not col_esp because all fifteen former Spanish-Portuguese colonies in the sample enter as majoritarian democracies.